https://ntrs.nasa.gov/search.jsp?R=19860016858 2020-03-20T14:22:53+00:00Z

W 0802171 ANX

AR 86-2

Aeronautical Report 86-2

THEORETICAL ANALYSIS OF THE ELECTRICAL ASPECTS OF THE BASIC ELECTRO-IMPULSE PROBLEM IN AIRCRAFT DE-ICING APPLICATIONS

ΒY

Robert A. Henderson and

Robert L. Schrag

Aeronautical Engineering Department Wichita State University Wichita, Kansas 67208

Technical Report

for

NASA Lewis Research Center (NAG-3-284) NASA Technical Monitor: J. J. Reimann

June 1986

(NASA-CR-176808) THEORETICAL ANALYSIS OF	N86-26330	26330	
THE ELECTRICAL ASPECTS OF THE EASIC			
ELECTRO-IMPULSE PROBLEM IN AIRCRAFT DE-ICING	·		
APPLICATIONS (Wichita State Univ.) = 108 p	Unclas		
BC A06/MF A01 CSCL 01C G3/05	43454 881-		

Preface

-4

This report is a part of the research and development project on aircraft de-icing by the electromagnetic impulse method. This project has been sponsored by the Lewis Research Center of the National Aeronautics and Space Administration under grant number NAG-3-284. The grant administrator was Mr. John J. Reinmann. For the previous four years, many tests had been run in the NASA Icing Research Tunnel and three sets of flight tests were performed. These, plus various laboratory tests, have resulted in a semi-empirical technology for designing an electro-impulse de-icing (EIDI) system.

However, the empirical method is inadequate when a very different geometry, material or size is encountered. A computative solution is needed which permits prediction of the de-icing effect for a given configuration and electrical circuitry. This report, which is principally the Ph.D. dissertation of Robert A. Henderson under the direction of Prof. Robert L. Schrag, attempts to do the first part of a full computer simulation of EIDI. The pressure/time produced by the method in this report would be necessary input for a computer code giving the structural dynamic response of a given configuration. The configurations in mind are leading edge portions of aircraft wings, engine nacelle inlets and rotor blades. Applications, however, are not limited to these applications.

The author acknowledges the assistance of NASA-Lewis Research Center, both for the support of the whole EIDI project at Wichita State University and for the opportunity to work at NASA-Lewis during the summer of 1985. The assistance of Drs. R. Joseph Shaw, Bill Ford and Avram Sidi during that time is gratefully expressed.

TABLE OF CONTENTS

PAGE

4

- er

1.	INTRODUCTION						
	1.1	Electro-Impulse Phenomenon and De-Icing	1				
	1.2	Scope of Dissertation	З				
	1.3	Organization of Dissertation	Э				
2.	THE	HE ELECTRIC CIRCUIT MODEL					
з.	. TRANSMISSION LINE MODEL OF THE FIELD PROBLEM						
	3.1	Introduction	12				
	3.2	Geometry of a Prototype EIDI System	12				
	э. э	The Field Equations	14				
	3.4	Development of the Transmission Line Model	16				
4.	ANALYSIS STRUCTURE						
	4.1	Introduction	22				
	4.2	Method of Calculating Coil Impedance	22				
	4.3	Radial Magnetic Induction and Azimuthal Electric					
		Intensity	25				
		4.3.1 Target Surface Facing Coil	25				
		4.3.2 Target Surface Upposite Coll	30				
	4.4	Calculation of the Axial magnetic field Using the	~•				
		Iransmission Line Model	16				
	4.5	Force between larget and toll	32				
	4.0	Impulse Delivered to larget	33				
5.	A SPECIFIC SYSTEM EXAMPLE, INCLUDING EXPERIMENTAL RESULTS						
	5.1	Definition of the System	34				
	5.2	Coil Impedance Measurements	36				
	5.3	Current Waveform	36				
	5.4	Magnetic Field Measurements	37				
	5.5	Measurement of Impulse to Target	41				
6.	Comp	UTER ANALYSIS ON THE SYSTEM EXAMPLE	42				
·	6.1	Introduction	42				
	6.2	Coil Impedance	42				
	6.3 Current Waveform		45				
		6.3.1 Introduction	45				
		6.3.2 Current Before Clamp Diode Conduction	47				
		6.3.3 Current After Clamp Diode Conduction	50				
		6.3.4 Combining Pre- and Post Clamp Diode Conduction					
		Current	52				
	6.4	Magnetic Induction	53				

		6.4.1 Introduction	53
	(5.4.2 Radial Magnetic Induction on the Coil Side Face	
		of the Target	54
	(6.4.3 Axial Magnetic Induction on the Coil Side Face	
		of the Target	58
-	1	6.4.4 Magnetic Induction on the Opposite Face of the	
		Target	60
•	6.5	Force Versus Time	62
· ·	6.6	Impulse	65
•	7. CONCL	USION AND RECOMMENDATIONS FOR FURTHER WORK	67
	7.1	Conclusion	67
	•	7.1.1 Summary	67
-	•	7.1.2 Contributions by the Author	68
	7.2	Recommendations for Future Work	6 9
	APPENDIX	A: FORTRAN SOURCE CODE FOR CALCULATING COIL RESISTANCE	
		INCREASE DUE TO TARGET	72
	APPENDIX	B: FURIRAN SOURCE CODE FUR CALCULATING COTE CURRENT	36
		BEFURE CLAMP DIDDE CONDUCTION	/6
:	ADDENDTY	r_{1} calculation of $\kappa'(\lambda)$	70
	AFFENDIX	C: CALCULATION OF A (A)	/3
	APPENDIX	DE FORTRAN SOURCE CODE FOR CALCULATING RADIAL TIME	
·		DOMAIN MAGNETIC INDUCTION ON THE COIL SIDE SURFACE	
•		OF THE TARGET	83
			00
· ·	APPENDIX	E: FORTRAN SOURCE CODE FOR FORCE CALCULATION AT TIME	
· .		T=50 MICROSECONDS	95
	APPENDIX	F: NOTATION AND LIST OF SYMBOLS	9 7
:	REFERENCE	S	102
	VITA		105

-**4** . -

CHAPTER ONE

INTRODUCTION

1.1 ELECTRO-IMPULSE PHENOMENON AND DE-ICING

In a paper presented January 8, 1986, at the 24th AIAA Aerospace Sciences Meeting in Reno, Nevada, [1] author R. D. Rudich cited aircraft icing as the direct cause of four of the thirty two weather related fatal accidents reported in the paper. While this may not seem like a significant number, the loss of human lives in these four accidents could have been prevented if the aircraft involved had been able to cope successfully with the icing conditions they encountered.

The purpose of this dissertation is to present methods of analyzing the electromagnetic aspects of a new method of de-icing both private and commercial aircraft. This new de-icing method, referred to as Electro Impulse De-Icing (abbreviated EIDI), holds the promise of being superior to the present aircraft de-icing methods in terms of the energy expended in removing accreted ice [42].

The EIDI concept is not new. In May 1939, Great Britain issued a patent to Mr. Rudolf Goldschmidt covering the basic EIDI mechanism [2]. However, no commercial development of an EIDI system in the free world proceeded from this patent. It has only been within the last 5 years that the unavailability of bleed air from the new high bypass ratio engines used on the next generation commercial aircraft has caused attention to again be directed to the commercial use of an EIDI

system for removing ice from aircraft.

The simplest practical EIDI system consists of a spirally wound coil of rectangular cross section conductor mounted with its axis of symmetry perpendicular to the metal surface to be de-iced. An initially charged capacitor is discharged through this coil, and the resulting magnetic field from the coil's current causes eddy currents in the metal. The force exerted on these eddy currents by the coil's magnetic field is initially in such a direction as to cause the coil and the metal surface to separate. It is this force that causes ice on the metal surface to crack, and subsequently to be removed from the surface.

A considerable body of literature concerned with the electromagnetic aspects of a coil placed next to a conducting surface has accumulated. Levy [8] and Grover [9] present methods of calculating terminal impedances. Dodd and Deeds [10] discuss both impedance calculations and eddy current distributions. Once [11] is apparently the first researcher to apply a Hankel transformation in the calculation of impedances. Once's method is further developed and extended by El-Markabi and Freeman [4] to include calculation of the force between the coil and conductor when the coil current is a sinusoid. Bowley et. al. [43] discuss the use of the magnetic vector potential in calculating the impulse delivered to the conductor by a transient current in the coil. Lewis [44] summarizes the use of the Bowley et. al. method for designing a coil to deliver a specific impulse in an electro-impulse de-icing installation.

1.2 SCOPE OF DISSERTATION

An experimental set-up of a prototype EIDI system was assembled at The Wichita State University by Dr. Robert Schrag. An experimental study of the electro-impulse phenomenon in this system was made by Dr. Schrag utilizing field diagnostics methods [3]. Axial and radial components of the magnetic field were measured on both sides of the rigidly held aluminum "target" plate. The data were then used to deduce the total mechanical force versus time, and the mechanical impulse strength. Impulse strength was also measured directly with a ballistic pendulum. This experimental study provided results which will be used to verify theoretical predictions from the mathematical model used in this dissertation.

In this dissertation, the physical phenomena involved in the prototype EIDI system of Dr. Schrag's experiment will be investigated analytically and numerically. Specifically, the following tasks will be undertaken.

1. A mathematical model for the total electrical problem will be devised. It will employ a transmission line analogy to handle the electromagnetic field portion of the system, and a frequency domain model for the circuit portion.

2. The math model will be solved by computer for the specific set of conditions that existed in Dr. Schrag's diagnostics experiment, and the calculated and experimental results will be compared.

1.3 ORGANIZATION OF DISSERTATION

Figure 1 summarizes the analysis structure developed in this dissertation to predict the behavior of the prototype EIDI system briefly described in Section 1.1. Each of the blocks in this figure



FIGURE 1 Analysis Flow Diagram

-

represents a stage in the procedure for determining the force-time profile on a rigid coil placed next to a fixed conducting plane when a capacitor is discharged through the coil.

In Block 1 of Figure 1, Maxwell's equations are written for the coil and metal target of the physical system. A simple model of the coil is proposed that stresses the geometrical symmetry of the system, eliminating several terms from the Maxwell equations. Chapter 3 discusses this modeling of the coil and metal target, and shows that application of a Hankel transform to the model equations results in a conceptual replacement of the field problem with an infinite set of transmission line problems (Block 2).

In Block 3 of Figure 1, the analysis of the transmission line model is performed. The methods of this analysis, for steady state sinusoidal conditions, may be found in Chapter 4. Chapters 3 and 4 provide the basic theoretical developments, with the results of the analyses in Chapter 4 applied in Chapter 6 to Dr. Schrag's prototype experimental EIDI system, described in Chapter 5. Modeling of the circuit used to provide energy to the coil is discussed in Chapter 2 for the specific experimental system of Chapter 5.

Once the transmission line model of the coil and target has been established in Block 3, Figure 1 shows two possible next steps in the analysis. One of these next steps, calculation of the true field values (Block 7), requires a knowledge of the coil current. Accordingly, one proceeds from the transmission line analysis in Block 3 to the Block 4 calculation of the coil impedance, required for calculating the coil current. Use of the transmission line model for calculating the part of the impedance of the coil that is due to the

interaction between the coil and target is developed theoretically in Section 4.2 of Chapter 4, and applied to the impedance calculation of the coil in the prototype EIDI system in Section 6.2 of Chapter 6.

Coil impedance obtained in Block 4 is added to the calculated skin effect loss resistance in the coil, and with this total coil impedance the problem of calculating coil current, using the circuit models presented in Chapter 2, is addressed. This is Block 5, discussed in Section 6.3 of Chapter 6 for the prototype EIDI system described in Chapter 5.

Knowledge of the frequency spectrum of the coil current allows the calculation of the true field values in Hankel-Fourier space, shown in Block 7 and developed theoretically in Section 4.3 of Chapter 4. Performing an inverse Hankel transformation on these fields results in Fourier space (frequency domain) fields. An inverse discrete Fourier transform applied to the Fourier space fields provides the time domain behavior of the electric and magnetic fields, shown in Block 8. This procedure is discussed in Section 6.4 of Chapter 6 for the prototype EIDI system.

Finally, the real space axial and radial magnetic induction fields on the coil side face of the metal plate provide the necessary information for calculating the total force on the target as a function of time, as shown in Block 9. Theoretical development of the force equation is in Section 4.5 of Chapter 4, with Section 6.5 of Chapter 6 describing the numerical implementation of this force calculation for the prototype EIDI system. Calculation of the impulse delivered to the target is discussed theoretically in Section 4.6 of Chapter 4, and its application to the prototype EIDI system is de-

scribed in Section 6.6 of Chapter 6.

Chapter 7 contains a summary of the results obtained, a description of the original contributions made by the author, and some suggestions for improving the procedures described in this dissertation for modeling the electrical aspects of an EIDI system.

CHAPTER TWO

THE ELECTRIC CIRCUIT MODEL

Modeling of the electronics providing power to the de-icing coil in an EIDI system is performed with the desired output from the model in mind. Since it is the coil current in conjunction with the physical configuration of the coil and wing skin that determines the magnetic fields responsible for the forces on the skin, the current in the circuit was chosen as the primary variable.

Testing of a prototype EIDI system began at The Wichita State University in 1982. The part of this system that provides power to the coil is shown in Figure 5A (page 35). This system was chosen as the basic physical configuration for which a circuit model would be derived. This circuit model is then analyzed to determine the shape of the coil current. A simple circuit model that is suggested by the physical system in Figure 5A is shown in Figure 6E (page 47). This is the basic circuit model, valid when the clamp diode across the capacitor is not conducting.

Justification for modeling the EIDI system as a lumped parameter circuit was provided by the experimental verification of the absence in the signals present of any significant frequency components having wavelengths comparable to the physical dimensions of the system. Because of the complex electromagnetic interaction between the coil and the skin next to it, it was felt that time domain modeling of the

coil's terminal v-i characteristics would be too complex to be of much use. Consequently, a frequency domain approach was chosen for analyzing the circuit, making the model of the coil a linear frequency dependent impedance.

The presence of the SCR and the clamp diode across the capacitor makes the circuit model nonlinear, so that straightforward Fourier (frequency domain) techniques are not applicable. This difficulty is circumvented by performing a piecewise linear analysis of the circuit. In this analysis, the physical circuit is modeled by one of two possible circuits, depending on the state of the clamp diode. Initially, when the SCR has just been triggered, the diode is assumed off (an open circuit) and the model of Figure 6E (page 47) is used for analysis. In addition to calculating the current in this circuit, the capacitor voltage is also calculated. When this voltage first becomes negative, the clamp diode is modeled as coming into immediate forward conduction: This diode then acts as a short circuit, resulting in the circuit model shown in Figure 6F (page 51). All subsequent (in time) circuit calculations are then performed with this model.

Theoretical justification for modeling both the SCR and the clamp diode as simple on-off switches is now provided. However, the ultimate justification for such an outright dismissal of the effects of both the SCR and the clamp diode in determining the current waveform comes from observing how closely the predicted current waveform (using the models that ignore the non-ideal nature of the SCR and diode) agrees with the measured waveform. This will be shown in Chapter 6.

In a practical EIDI system, the voltage on the energy storage

capacitor prior to its discharge is 1000 to 1500 volts. When the SCR is triggered into conduction, its voltage drop is on the order of 1 volt, which is small compared to the capacitor voltage, and so may be ignored in determining the circuit current. In addition to ignoring the voltage drop across the SCR, the dynamics of the SCR are also not modeled. Such dynamics are, in the EIDI circuit, primarily manifested in the failure of the SCR to trigger into instantaneous full forward conduction upon initiation of a forward gate current. However, modern SCR design techniques [5], [6], [7] have resulted in turn-on times that are short compared to the time required for a significant change in the coil current in the experimental EIDI prototype system. The SCR used in the prototype system had a di/dt rating of 800 amps/ microsecond, while the maximum observed rate of change in the circuit vas 15 amps/microsecond.

A similar consideration of the voltage levels in the circuit results in the conclusion that the approximately 1 volt forward drop across the clamp diode is not a primary factor in determining circuit current. When the diode is off, its transition capacitance is insignificant compared to the capacitance of the energy storage capacitor. Reverse leakage current in the diode is ignored due to the large energy storage capacitor in parallel with the diode and the relatively small time in which the electrical events of interest take place in the circuit. With the diode in forward conduction, the sum of its transition and diffusion capacitances are small enough that, to a first approximation, they may be ignored in the circuit model.

In constructing the EIDI prototype experimental configuration, care was exercised to minimize parasitics in the circuit. Special low

inductance and resistance cable was used to connect the energy storage capacitor to the coil. However, both of these cable parameters were measured in the prototype system and are taken into account in the circuit model. The energy storage capacitor, which was physically several units in parallel, used copper strap for wiring connections to minimize inductance and resistance parasitics. The ESR (equivalent series resistance) and ESL (equivalent series inductance) of the capacitors were felt to be small, and are not modeled. If, in a particular EIDI installation, these parasitics are not small, they can be taken into account by adding lumped elements in series with the energy storage capacitor in the circuit model.

Accurate frequency domain modeling of the coil is the most difficult feat in constructing the circuit model, and is discussed in Chapter 4.

CHAPTER THREE

TRANSMISSION LINE MODEL OF THE FIELD PROBLEM

3.1 INTRODUCTION

The transmission line model of the coil and metal plate is the heart of the prototype EIDI system model. It is this model that is expected to account for the complex electromagnetic interactions between the coil and plate. These interactions help establish the impedance presented at the coil terminals, and so are a factor in determining the coil current. This current is important in determining the force on the plate.

3.2 GEOMETRY OF A PROTOTYPE EIDI SYSTEM

Figure 3A shows the profile of the physical coil-metal plate con-



figuration we have chosen to model. The coil has the shape of a short (h<<R_g) thick walled ($R_1 << R_g$) hollow cylinder with an inside

radius R, and an outside radius R_a , whose axis is perpendicular to a flat metal plate (henceforth called the "target") of thickness d extending to infinity in all radial directions. An air gap exists between the coil and target, both of which are assumed rigid and stationary in space.

Most of the coils used in EIDI applications have been round, at least prior to a possible bending of the coil to conform to the curved leading edge of a wing. The flat geometry of Figure 3A is a reasonable model of such a coil, provided that the radius of curvature of the coil, after being shaped to conform to the wing, is much greater than the coil outer radius R_e .

The assumption that the coil and target remain separated by a fixed distance would be reasonable if the initial separation distance was much larger than the maximum change in separation distance obtained when the capacitor is discharged through the coil. Such an inequality in separation distances may not hold in a practical EIDI installation. Alternatively, the fixed separation distance assumption would be justified if the force "impulse" delivered to the plate by the coil was so short that the plate acquired only a small velocity, with negligible displacement, for the duration of the "impulse". This does not generally happen in a practical EIDI system. In fact, a well designed installation has the target move from zero to maximum displacement within the duration of the "impulse". With ice loading present, however, the displacement may be small compared to the initial separation distance.

3.3 THE FIELD EQUATIONS

The approach that appears to be the most fruitful for modeling the coil and target in our prototype EIDI system is presented in a paper by El-Markabi and Freeman [4]. We now describe their approach, emphasizing those aspects of the theory that are most appropriate to the analysis of the model of the prototype system of Figure 3A. The reader is referred to El-Markabi and Freeman [4] for the more general theory.

Current in the coil of Figure 3A is assumed to be entirely phi directed. A slight modeling error is introduced by this assumption, since a radial component of current actually exists in a practical coil due to the spiral winding of the coil. By neglecting this small radial current, a model having azimuthal symmetry is obtained. Consequently, the model shows no phi dependence in any of its field quantities, the electric field contains only a phi component, and the magnetic field contains only axial and radial components.

Because most of the spectral energy of the current in a practical EIDI coil is confined to relatively low frequencies, a quasi-static situation is assumed. The displacement current term in Ampere's Law is ignored.

With these assumptions, Maxwell's equations written for Figure 3A become

$$\frac{\partial E_{\phi}(z,r,t)}{\partial z} = \mu \frac{\partial H_{r}(z,r,t)}{\partial t}$$

$$\frac{\partial H_{r}(z,r,t)}{\partial r} = -\mu \frac{\partial H_{z}(z,r,t)}{\partial t}$$

$$\frac{\partial H_{r}(z,r,t)}{\partial z} - \frac{\partial H_{z}(z,r,t)}{\partial r} = \sigma E_{\phi}(z,r,t)$$

Partial differentiation with respect to time can be eliminated from these equations by taking the Fourier transform. Differentiation with respect to time is then replaced with multiplication by the $j^{(\omega)}$ operator. Performing this transformation, we obtain the following equations (note that we have not introduced new symbols for the transformed field quantities, but have instead simply replaced their time variable argument t with the Fourier transform variable omega)

$$\frac{\partial E_{\phi}(z,r,\omega)}{\partial z} = j\omega\mu H_{r}(z,r,\omega) \qquad (1)$$

$$\frac{1}{r} \frac{\partial}{\partial r} \left[r E_{\phi}(z, r, \omega) \right] = -j \omega \mu H_{z}(z, r, \omega)$$
(2)

$$\frac{\partial H_r(z,r,\omega)}{\partial z} - \frac{\partial H_z(z,r,\omega)}{\partial r} = \sigma E_{\phi}(z,r,\omega)$$
(3)

Equation (1) contains only the phi component of E and the radial component of H. By combining equations (2) and (3) in such a fashion as to eliminate the axial component of H, we would have another equation containing only the phi component of E and the radial component of H. This elimination yields

$$\frac{\partial H_r}{\partial z} = \sigma E_{\phi} + \frac{\partial H_z}{\partial r} = \sigma E_{\phi} + \frac{\partial}{\partial r} \left[\frac{-1}{j \omega \mu} \frac{1}{r} \frac{\partial}{\partial r} (r E_{\phi}) \right]$$

$$\frac{\partial H_r}{\partial z} = \sigma E_{\phi} - \frac{1}{j \omega \mu} \frac{\partial}{\partial r} \left[\frac{1}{r} \frac{\partial}{\partial r} (r E_{\phi}) \right] \qquad (4)$$

At this point, we introduce a mathematical tool that is of considerable use here. This tool is the Hankel transform of order n, defined by [12]

$$\mathcal{H}{f(r)} \triangleq \int_{0}^{\infty} f(r) r J_{n}(\lambda r) dr = F(\lambda)$$

with the inverse transform

$$4^{-1}\{F(\lambda)\} = \int_{0}^{\infty} F(\lambda) \lambda J_{n}(\lambda r) d\lambda = f(r)$$

Although the introduction of an additional transform introduces difficulties of its own, it provides an even greater amount of simplification, and makes possible the transmission line model of the coiltarget electromagnetic field problem.

If a Hankel transform of order 1 is applied with respect to the variable r in equation (4), the following result is obtained [11], [12] (note that we have again not introduced new symbols for the transformed quantities, simply replacing their spatial variable argument r with the Hankel transform variable lambda)

$$\frac{\partial H_{r}(z,\lambda,\omega)}{\partial z} = (\sigma + \frac{\lambda^{2}}{j\omega\mu}) E_{\mu}(z,\lambda,\omega)$$
 (5)

The advantage of equation (5) over equation (4) is that equation (5) contains partial derivatives with respect to only the z coordinate, in effect becoming an ordinary differential equation (if one allows "constants of integration" for such an equation to be arbitrary functions of all variables except z). If the same Hankel transform is applied to equation (1), we obtain

$$\frac{\partial E_{\phi}(z,\lambda,\omega)}{\partial z} = j\omega\mu H_{r}(z,\lambda,\omega)$$
(6)

Equations (5) and (6) are two coupled ordinary differential equations that are recognizable as the canonical transmission line equations. The solutions of these equations are well known (13), (14), (15).

3.4 DEVELOPMENT OF THE TRANSMISSION LINE MODEL

Because equations (5) and (6) are identical in form to the equations describing voltage and current on an ordinary transmission line, we now introduce the symbols \widehat{V} and $\widehat{\Gamma}$ (each of which is a function of position z along the coil axis, the Fourier variable omega, and the Hankel variable lambda) to represent the phi component of E and the radial component of H respectively. Then equations (5) and (6) become

$$\frac{\partial \widehat{T}}{\partial z} = (\sigma + \frac{\lambda^2}{j\omega\mu}) \widehat{V}$$
(7)
$$\frac{\partial \widehat{V}}{\partial z} = j\omega\mu \widehat{T}$$
(8)

Differentiating equation (7) with respect to z, we have

$$\frac{\partial^{2} \hat{T}}{\partial z^{2}} = (\sigma + \frac{\lambda^{2}}{j\omega\mu}) \frac{\partial \hat{V}}{\partial z}$$

Substitution of equation (8) into this last equation yields

$$\frac{\partial^{2} \hat{\Gamma}}{\partial z^{2}} = j \omega \mu \left(\tau + \frac{\lambda^{2}}{j \omega \mu} \right) \hat{\Gamma}$$

$$\frac{\partial^{2} \hat{\Gamma}}{\partial z^{2}} = \left(\lambda^{2} + j \omega \mu \tau \right) \hat{\Gamma}$$
(9)

Similarly, we differentiate equation (8) to obtain

 $\frac{\partial^2 \widehat{V}}{\partial z^2} = j \omega_{\mu} \frac{\partial \widehat{T}}{\partial z}$

Substitution of equation (7) into this equation yields

$$\frac{\partial^{2} \hat{V}}{\partial z^{2}} = j \omega \mu \left(\sigma + \frac{\lambda^{2}}{j \omega \mu} \right) \hat{V}$$

$$\frac{\partial^{2} \hat{V}}{\partial z^{2}} = \left(\lambda^{2} + j \omega \mu \sigma \right) \hat{V} \qquad (10)$$

Now define the complex propagation constant gamma in Fourier-Hankel space as

$$\chi = \sqrt{\lambda^2 + j\omega\mu\sigma}$$

so that the general solutions to equations (9) and (10), assuming complex sinusoidal time variation of \widehat{V} and \widehat{I} , may be written as

$$\hat{T} = \hat{T}_{o+} e^{-\lambda z} + \hat{T}_{o-} e^{\lambda z}$$
 (11)

$$\widehat{\mathbf{V}} = \widehat{\mathbf{V}}_{0+} e^{-\widetilde{\mathbf{V}}_{0+}} + \widehat{\mathbf{V}}_{0-} e^{\widetilde{\mathbf{V}}_{0-}}$$
(12)

where $\widehat{\mathbf{I}}_{0..}$, $\widehat{\mathbf{I}}_{0..}$ and $\widehat{\mathbf{V}}_{0..}$, $\widehat{\mathbf{V}}_{0..}$ are complex phasor functions of the Hankel variable lambda (we have suppressed the complex sinusoidal time variation $e^{\hat{\mathbf{j}}^{tot}}$ in these solutions).

Note that we are conceptually dealing with an uncountably infinite set of transmission lines. Each transmission line is associated with a different value of lambda, where lambda is non-negative. The physical origin of this infinite number of transmission lines lies in our "compressing" the infinite radial variation in the real space field quantities H, and E_{ϕ} at a given axial coordinate z into variables V and I localized to a single "point" (the corresponding z coordinate) on a Hankel space transmission line. It then takes an infinite number of transmission lines to account for the infinite number of possible values of the radius in the real space problem.

To derive an expression for the characteristic impedance of one of these Hankel space transmission lines, consider the case of a line having only a single frequency complex sinusoid traveling in the direction of increasing z. In agreement with ordinary transmission line theory, the negative sign in the exponents in equations (11) and (12) denotes propagation in the direction of increasing z. Then equations (11) and (12) become

$$\hat{I} = \hat{I}_{0+} e^{-\chi_{z}}$$
(13)

$$\hat{V} = \hat{V}_{0+} e^{-8E} \qquad (14)$$

Substitute the expressions for the current and voltage from equations (13) and (14) into equation (7)

$$-I_{0+} \chi e^{-\chi z} = (\sigma + \frac{\lambda^2}{j\omega\mu}) V_{0+} e^{-\chi z}$$

From this equation, we obtain the defining expression for the characteristic impedance of a Hankel space transmission line as

$$Z_{o} \triangleq \frac{V_{o+}}{I_{o+}} = -8 \left(\sigma + \frac{\lambda^{2}}{j\omega\mu}\right)$$
$$Z_{o} = \frac{-j\omega\mu}{\sqrt{\lambda^{2} + j\omega\mu\sigma}}$$

This result differs by a negative sign from the expression given in El-Markabi and Freeman [4] for the line's characteristic impedance. One could argue that the negative branch of the square root function in the expression for the propagation coefficient gamma should be chosen, which would eliminate the negative sign in the characteristic impedance. However, calculations performed with a negative characteristic impedance results in predictions that are essentially in agreement with experimental measurements, whereas the use of a positive characteristic impedance predicts results that are not in agreement with experiment. Alternatively, one could propose that it is the positive sign that must be chosen for the exponents in equations (11) and (12) to correspond to propagation in the direction of increasing z. Such an assumption also results in predictions that are not in agreement with experiment.

Transformation of real space current sources into Hankel space is discussed in El-Markabi and Freeman (4). They show that a disc of azimuthally directed uniform surface current of phasor value \widetilde{I} , with an inner radius R₁ and an outer radius R₂, located on the z-axis, becomes a sinusoidal current source with phasor value

$$\frac{\widetilde{I}}{R_2 - R_1} \int_{R_1}^{R_2} - J_1(\lambda r) dr$$

connected in parallel with the Hankel space transmission line. The z coordinate of this current source is the same as the z coordinate of the real space surface current from which it arose.

We have chosen three of these ideal azimuthally directed discs of uniform surface current to model the current distribution in the coil. The inner (outer) radius of the disc is equal to the inner (outer) radius of the coil. The middle disc is located at the axial center of the coil, while the two outer discs are each spaced out a distance of one-third of the coil's width from the center of the coil. See Figure 38. Clearly, some error occurs here due to the localization of the model's current to these discs.



FIGURE 3B CURRENT DISC MODEL OF COIL

Figure 3C shows the Hankel space transmission line configuration corresponding to the three current sheet model of the coil next to the target. All of our Hankel space calculations are based upon this model.





CHAPTER FOUR

ANALYSIS STRUCTURE

4.1 INTRODUCTION

In this chapter, the transmission line model of the coil and target developed in Chapter 3 and shown in Figure 3C is examined mathematically using conventional transmission line theory to predict the Fourier-Hankel space behavior of the model when it is in the sinusoidal steady state condition. The analyses performed in this chapter will assume each of the three current sources in Figure 3C has unit phasor value. In view of the analogy developed in Chapter 3, we will freely use the terms current and voltage in connection with the transmission line model to stand for the Fourier-Hankel transforms of the radial magnetic induction B, (z,r,t) and the azimuthal electric intensity $E_{\phi}(z,r,t)$ respectively. Throughout this chapter, reference should be made to Figure 3C for insight into the equations written and for identification of the variables used.

4.2 METHOD OF CALCULATING COIL IMPEDANCE

Although the part of the coil impedance that is independent of the properties of the ribbon conductor used to wind the coil is calculated in Hankel space, as shown in this section, one must begin the derivation of the expression for the impedance in Fourier space. Only in the frequency domain does the concept of impedance make sense. We will calculate the coil impedance at a frequency omega by exciting the

coil with a current source of phasor value 1, and finding the phasor voltage response at the coil terminals. The coil impedance is then numerically equal to this voltage response.

Our real (Fourier) space model of the coil is three ideal discs of azimuthally directed uniform surface current. We need to relate the electric fields induced in the different parts of this model to the single voltage that exists between the coil terminals. We approximate the voltage appearing at the coil terminals by the average of the voltages induced at the locations of the three discs of the model. Symbolically, we have

$$Z = \frac{\widetilde{V}}{\widetilde{T}} = \frac{\widetilde{V}}{1} = \widetilde{V} = \frac{1}{3} \left(\widetilde{V}_{0} + \widetilde{V}_{1} + \widetilde{V}_{2} \right)$$

where $\widetilde{V_k}$ represents the voltage induced at the location of the disc with axial coordinate z_k .

Because of the symmetry of our model, the voltage induced on a circular path (centered on the z axis) of radius r and axial coordinate z_k will be given by

$$2\pi r E_{\phi}(z_k, r, \omega)$$

The average voltage induced on the infinite number of circular paths between $r=R_1$ and $r=R_2$ is

$$\frac{1}{R_2-R_1}\int_{R_1}^{K_2} 2\pi r E_{\phi}(z_k, r, \omega) dr$$

Multiplication of this expression by N, the number of turns in the actual coil, yields

$$\widetilde{V}_{k} = \frac{a\pi N}{R_{2} - R_{1}} \int_{R_{1}}^{R_{2}} r E_{\phi}(z_{k}, r, \omega) dr$$

which is the model-predicted voltage induced on an infinitesimally thin N turn coil having an inner radius R_1 and an outer radius R_2 located at z_* . Averaging these induced voltages over the three discs results in the phasor coil voltage

$$\widetilde{V} = \frac{1}{3} \left\{ \frac{2\pi N}{R_2 - R_1} \int_{R_1}^{R_2} r \stackrel{2}{\underset{i=0}{\overset{\sim}{\underset{i=0}{\overset{\scriptstyle}{\underset{i=0}{\overset{\scriptstyle}{\underset{i=0}{\overset{\scriptstyle}{\underset{i=0}{\overset{\scriptstyle}{\underset{i=0}{\underset{i=0}{\overset{\scriptstyle}{\underset{i=0}{\underset{i=0}{\overset{\scriptstyle}{\underset{i=0}{\overset{\scriptstyle}{\underset{i=0}{\overset{\scriptstyle}{\underset{i=0}{\underset{i=0}{\overset{\scriptstyle}{\underset{i=0}{\underset{i=0}{\overset{\scriptstyle}{\underset{i=0}{\underset{i=0}{\underset{i=0}{\overset{\scriptstyle}{\underset{i=0}{\underset{i=0}{\overset{\scriptstyle}{\underset{i=0}{\atopi=0}{\underset{i=0}{\underset{i=0}{\underset{i=0}{\underset{i=0}{\underset{i=0}{\underset{i=0}{\atopi=0}{\underset{i=0}{\atopi=0}{\underset{i=0}{\atopi=0}{\underset{i=0}{\atopi=0}{\underset{i=0}{\atopi=0}{\underset{i=0}{\atopi=0}{\atopi=0}{\underset{i=0}{\atopi=$$

Because this expression is numerically equal to the coil impedance, we will replace the symbol \widetilde{V} by Z in further appearances of this expression.

The E fields that appear in (1) are frequency domain (Fourier space) E fields. In terms of the Fourier-Hankel space E fields, we can write these Fourier space E fields as

$$E_{\phi}(z_{k},r,\omega) = \int_{0}^{\infty} E_{\phi}(z_{k},\lambda,\omega) \lambda J_{\mu}(\lambda r) d\lambda \qquad (2)$$

where we have again used the list of arguments to distinguish between different functions. Specifically, $E_{\phi}(z_k, r, \omega)$ is the Fourier transform of the real space azimuthal electric intensity $E_{\phi}(z_k, r, t)$, and $E_{\phi}(z_k, \lambda, \omega)$ is the Fourier-Hankel transform of the real space azimuthal electric intensity $E_{\phi}(z_k, r, t)$.

$$Z = \frac{1}{3} \frac{2\pi N}{R_2 - R_1} \int_{R_1}^{R_2} \left[\sum_{i=0}^{2} \int_{0}^{\infty} E_{\phi}(z_i, \lambda, \omega) \lambda J_i(\lambda r) d\lambda \right] r dr$$

Interchanging the order of integration,

where

$$K'(\lambda) = \frac{N}{3(R_2 - R_1)} \int_{R_1}^{R_2} J_1(\lambda r) r dr$$

is the Hankel transformed current corresponding to an infinitesimally thin N turn coil of inner radius R_1 and outer radius R_2 , carrying a phasor current of strength 1/3.

Expression (3) contains three Fourier-Hankel space electric fields (transmission line voltages) that conceptually arose from Fourier space phasor currents of value 1/3 on each of the three discs in Figure 3B. The "scaling factor" that relates Fourier-Hankel space fields due to unity Fourier space phasor coil current to Fourier-Hankel space voltages and currents due to unity phasor current in each of the three current sources in Figure 3C is $K'(\lambda)$. Then in terms of the Fourier-Hankel space voltage $E_{\phi}'(z_*, \lambda, \omega)$ due to unity phasor current in each of the Figure 3C current sources, (3) becomes

$$Z = 2\pi \int_{0}^{\infty} \left[K'(\lambda) \right]^{2} \stackrel{2}{\underset{i=0}{\overset{2}{\underset{i=0}{\atopi=0}{\overset{2}{\underset{i=0}{\overset{2}{\underset{i=0}{\overset{2}{\underset{i=0}{\atopi=0}{\overset{2}{\underset{i=0}{\atopi=0}{\underset{i=0}{\overset{2}{\underset{i=0}{\atopi=0}{\atopi=0}{\overset{2}{\underset{i=0}{\atopi=0}{\atopi=0}{\overset{2}{\underset{i=0}{\atopi=0}{\atopi=0}{\atopi=0}{\underset{i=0}{\atopi=0}$$

Expression (4) is the result that we use to calculate the terminal impedance of the coil.

4.3 <u>RADIAL MAGNETIC INDUCTION AND AZIMUTHAL ELECTRIC INTENSITY</u> 4.3.1 Target Surface Facing Coil

We begin by calculating the characteristic impedance and the complex propagation constant of the air transmission line and the metal transmission line, referred to hereafter as the air line and the metal line respectively, using the results derived in Chapter 3. For the air line, the conductivity or is equal to zero, and so

$$Z_a = -j\omega\mu / \sqrt{\lambda^2 + j\omega\mu\sigma} = -j\omega\mu / \lambda$$
$$X_a = \sqrt{\lambda^2 + j\omega\mu\sigma} = \lambda$$

For the metal line, no simplification is possible and we have

$$Z_{m} = \frac{-j\omega\mu}{\sqrt{\lambda^{2} + j\omega\mu\sigma}}$$
$$\delta_{m} = \sqrt{\lambda^{2} + j\omega\mu\sigma}$$

According to the analogy developed in Chapter 3, calculation of the total current and voltage at z=0 in Figure 3C is equivalent to calculation of the radial magnetic induction B, and the azimuthal electric intensity E_{φ} on the coil side face of the target. We perform this current and voltage calculation by replacing the transmission line configuration for all z>0 with its equivalent impedance. Note that both sides of the metal line are connected to infinite lengths of air transmission line with characteristic impedance Z. The equivalent impedance looking to the right at z=0 in Figure 3C is

$$Z(0) = Z_m \frac{Z_a + Z_m \tanh(X_m d)}{Z_m + Z_a \tanh(X_m d)}$$
(5)

Using this equivalent impedance, the current reflection coefficient at z=0 as seen from the air line is

$$P_{H} = \frac{Z_{a} - Z(0)}{Z_{a} + Z(0)}$$

and the total current at z=0 is then

$$\widehat{\mathbf{I}}_{i,o,i+1} = \widehat{\mathbf{I}}_{i,n,o} + \widehat{\mathbf{I}}_{r,o,i+1,o,i+1,o} = (1 + \rho_{H})\widehat{\mathbf{I}}_{i,n,o}$$
(6)

in terms of the incident current \widehat{I}_{inc} , which must be due solely to the three current sources. Expression (6) may be simplified as follows.

$$\hat{I}_{total} = \left(1 + \frac{Z_a - Z(o)}{Z_a + Z(o)}\right) \hat{I}_{inc} = \frac{2Z_a}{Z_a + Z(o)} \hat{I}_{inc} \quad (7)$$

Substituting (5) into (7) yields

$$\widehat{I}_{total} = 2\widehat{I}_{inc} \frac{Z_a}{Z_a + Z_m \tanh(X_m d)} \frac{Z_a + Z_m \tanh(X_m d)}{Z_m + Z_a \tanh(X_m d)}$$

which, upon replacing Z. and Z. by their defining expressions gives

$$\widehat{I}_{1,0,1,1} = 2\widehat{I}_{1,0,0} - \frac{1}{1 + \frac{\lambda^2}{\sqrt{\lambda^2 + j\omega\mu\sigma}}} \frac{1}{\tan h(8_md)}$$
(8)
$$1 + \frac{\lambda}{\sqrt{\lambda^2 + j\omega\mu\sigma}} - \frac{\lambda}{\sqrt{\lambda^2 + j\omega\mu\sigma}} + \tanh(8_md)$$

Now consider the current source in Figure 3C closest to the target. This source produces an incident current on the metal line at z=0 given by

$$\hat{J}_{inco} = \frac{1}{z} e^{-\chi_a} g = \frac{e^{-\lambda}g}{z}$$

where the factor 1/2 comes from the equal division of the unity phasor amplitude current to both sides of the air line containing the current source. Total current incident at z=0 is by superposition given by

$$\hat{T}_{inc} = \frac{1}{2} e^{-\lambda g} + \frac{1}{2} e^{-\lambda (g+h/3)} + \frac{1}{2} e^{-\lambda (g+2h/3)}$$

$$\hat{T}_{inc} = \frac{1}{2} e^{-\lambda g} \left(1 + e^{-\lambda h/3} + e^{-2\lambda h/3}\right) \qquad (9)$$

Substituting (9) into (8) yields

$$\widehat{I}_{1} = \frac{e^{-\lambda g} \left(1 + e^{-\lambda h/3} + e^{-2\lambda h/3}\right)}{1 + \frac{\lambda}{\sqrt{\lambda^2 + j\omega\mu\sigma}}} \qquad (10)$$

$$\frac{1 + \frac{\lambda}{\sqrt{\lambda^2 + j\omega\mu\sigma}}}{\frac{\lambda}{\sqrt{\lambda^2 + j\omega\mu\sigma}} + \tanh(\chi_m d)} \qquad (10)$$

for the total steady state current $\widehat{I}_{1,\ldots,1}$ at z=0 due to unit strength complex sinusoidal excitation in each of the three current sources. Multiplying (10) by K'(λ)I(ω), where I(ω) is the Fourier transform of the current in the coil (and multiplication by K'(λ) transforms a unity Fourier space current into the corresponding Fourier-Hankel space current), yields the desired result

$$\widehat{I}(z=0) = \frac{I(\omega) K'(\lambda) e^{-\lambda g} (1+e^{-\lambda h/3}+e^{-\lambda h/3})}{1+\sqrt{\lambda^2+j\omega\mu\sigma} tanh (\aleph_m d)}$$
(11)
$$\frac{1+\sqrt{\lambda^2+j\omega\mu\sigma}}{\sqrt{\lambda^2+j\omega\mu\sigma}} + tanh (\aleph_m d)$$

for the Fourier-Hankel transform of the radial magnetic induction B. on the coil side face of the metal target when the coil current spectrum is $I(\omega)$.

Calculation of the voltage is almost identical to the calculation of the current performed above. The voltage reflection coefficient is the negative of the current reflection coefficient,

$$P_{E} = \frac{Z(0) - Z_{a}}{Z(0) + Z_{a}}$$

Total voltage at z=0 is

 $\widehat{V}_{total} = \widehat{V}_{inc} + \widehat{V}_{rofloctod} = (1 + \rho_{E})\widehat{V}_{inc}$

in terms of the incident voltage $\widehat{V_{i,n}}$. Expression (12) may be simplified as

$$\widehat{V}_{total} = \left(1 + \frac{Z(0) - Z_a}{Z(0) + Z_a}\right) \widehat{V}_{inc} = \frac{2Z(0)}{Z_a + Z(0)} \widehat{V}_{inc} \quad (13)$$

Substituting (5) into (13),

$$\widehat{V}_{\text{total}} = 2\widehat{V}_{\text{inc}} \frac{Z_{\text{m}} + Z_{\text{m}} \tanh(\aleph_{\text{m}}d)}{Z_{\text{m}} + Z_{\text{a}} \tanh(\aleph_{\text{m}}d)}$$

$$\frac{Z_{\text{m}} + Z_{\text{m}} \tanh(\aleph_{\text{m}}d)}{Z_{\text{m}} + Z_{\text{m}} \tanh(\aleph_{\text{m}}d)}$$

which, upon replacing Z, and Z. by their defining expressions, gives

Total incident voltage V_{inc} at z=0 is given by

$$\hat{\mathbf{V}}_{ine} = \mathbf{Z}_{e} \hat{\mathbf{I}}_{ine} = \frac{j\omega\mu}{\lambda} \frac{1}{a} e^{-\lambda a} (1 + e^{-\lambda h/3} + e^{-\lambda h/3})$$

using (9) for $\widehat{I}_{i,n}$. Substituting this expression for $\widehat{V}_{i,n}$ into (14) yields

$$\hat{V}_{\text{total}} = \frac{\frac{j\omega\mu}{\lambda} e^{-\lambda g} \left(1 + e^{-\lambda h/3} + e^{-2\lambda h/3}\right)}{1 + \frac{\lambda}{\lambda^2 + j\omega\mu\sigma} + \frac{\lambda}{\sqrt{\lambda^2 + j\omega\mu\sigma}} + \frac{\lambda}{\sqrt{\lambda^2 + j\omega\mu\sigma}} \left(15\right)}$$
(15)

for the total voltage at z=0 due to unit strength complex sinusoidal excitation in each of the three current sources. Multiplying (15) by $K'(\lambda)I(\omega)$ yields the desired result

$$\widehat{\nabla}(z=0) = \frac{\mathbb{I}(\omega) \, \mathsf{K}'(\lambda) \, \frac{j\omega\mu}{\lambda} \, e^{-\lambda}g \left(1 + e^{-\lambda h/3} + e^{-2\lambda h/3}\right)}{1 + \frac{\sqrt{\lambda^2 + j\omega\mu\sigma}}{\lambda} \, \frac{\lambda}{1 + \sqrt{\lambda^2 + j\omega\mu\sigma} \, tanh(\mathbb{V}_m d)}}$$
(16)

for the Fourier-Hankel transform of the azimuthal electric intensity E_{ϕ} on the coil side face of the target when the coil current spectrum is $I(\omega)$.

4.3.2 Target Surface Opposite Coil

Knowing the total voltage and current at z=0 (derived in Section 4.3.1 and given by (16) and (11) respectively) allows a simple transmission line inverse chain matrix calculation of the total voltage and current at z=d, which are the Fourier-Hankel transforms of the azimuthal electric intensity E_{ϕ} and the radial magnetic induction B, on the surface of the metal target opposite the coil. Denoting these transforms of the fields on the coil side face of the target by \hat{V}_{*} (given by (16) above) and \hat{I}_{*} (given by (11) above), we have

$$\hat{V}_{f} = \hat{V}_{n} \cosh(\aleph_{m} d) - Z_{m} \hat{T}_{n} \sinh(\aleph_{m} d)$$
 (17)

$$\widehat{T}_{f} = -\widehat{V}_{n} \frac{\sinh(\aleph_{n}d)}{\mathbb{Z}_{m}} + \widehat{T}_{n} \cosh(\aleph_{m}d) \qquad (18)$$

where $\widehat{V_r}$ and $\widehat{I_r}$ denote total voltage (electric intensity) and current (magnetic induction) on the surface of the target opposite the coil.

4.4 <u>CALCULATION OF THE AXIAL MAGNETIC FIELD USING THE</u> TRANSMISSION LINE MODEL

The procedure that we use for calculating the force on the target is described in the next section, and requires both the radial and the axial components of the magnetic induction on the target surface next to the coil. The radial component of the magnetic intensity is available from the transmission line model by performing inverse Hankel and Fourier transformations on the calculated transmission line current at z=0, as discussed in Section 4.3.1. By using equation (2) in Chapter 3, it is also possible to calculate the axial component of the magnetic intensity (which does not have a transmission line analog) from the transmission line voltage, as shown below.

The azimuthal electric field is given in Fourier space by

$$E_{\phi}(z,r,\omega) = \mathcal{A}^{-1}\left\{V(z,\lambda,\omega)\right\} = \int_{0}^{\infty} V(z,\lambda,\omega) \lambda J_{1}(\lambda r) d\lambda \quad (19)$$

Substituting (19) into equation (2) from Chapter 3, we have

$$H_{z}(z,r,\omega) = \frac{-1}{j\omega\mu} \frac{1}{r} \frac{\partial}{\partial r} \left\{ r E_{\varphi}(z,r,\omega) \right\}$$

$$H_{z}(z,r,\omega) = \frac{-1}{j\omega\mu} \frac{\partial}{\partial r} \left\{ r \int_{0}^{\infty} \sqrt{(z,\lambda,\omega)} \lambda J_{1}(\lambda r) d\lambda \right\} (20)$$

Interchanging the order of partial differentiation and integration in (20),

$$H_{z}(z,r,\omega) = \frac{-1}{j\omega\mu r} \int_{0}^{\infty} \frac{\partial}{\partial r} \left\{ \lambda r J_{1}(\lambda r) \right\} V(z,\lambda,\omega) d\lambda \quad (21)$$

Using the identity

$$\frac{d}{dx} \left\{ x J_{1}(x) \right\} = x J_{0}(x)$$

(21) becomes

$$H_{z}(z,r,\omega) = \frac{-1}{j\omega\mu} \int_{0}^{\infty} V(z,\lambda,\omega) \lambda^{2} J_{0}(\lambda r) d\lambda \qquad (22)$$

Equation (22) states that the axial component of the magnetic intensity H. in Fourier space is given by the zero order inverse Hankel transform of the transmission line voltage at the corresponding z coordinate multiplied by lambda, with the result divided by $j\omega\mu$. Inverse Fourier transformation of the right hand side of (22) yields the desired time domain axial magnetic intensity.

4.5 FORCE BETWEEN TARGET AND COIL

Our procedure for calculating the total force between the target and the coil utilizes the Maxwell stress tensor, and is performed in real space (using the time domain fields). Stratton [41] shows that the total force F transmitted by a time varying electromagnetic field across a closed surface S is given by

$$\overline{F} = \oint \left[\varepsilon (\overline{\epsilon} \cdot \widehat{n}) \overline{\epsilon} + \frac{1}{\mu} (\overline{\epsilon} \cdot \widehat{n}) \overline{\epsilon} - \frac{1}{2} (\varepsilon \overline{\epsilon}^2 + \frac{\overline{\beta}^2}{\mu}) \widehat{n} \right] da$$

where \hat{n} is a unit vector normal to the surface. We take as our closed surface the plane z=0 (the coil side face of the target, "closed" at infinity). Since $\hat{n}=\hat{z}$, this integral reduces to

$$\overline{F} = \int_{r=0}^{\infty} \int_{\phi=0}^{2\pi} \left[\frac{B_2}{\mu} (B_2 \hat{z} + B_r \hat{r}) - \frac{1}{2} (\varepsilon E_{\phi}^2 + \frac{B_2^2 + B_r^2}{\mu}) \hat{z} \right] r d\phi dr$$

The total force tending to separate coil and target is just the z component of this force,

$$F_{2} = 2\pi \int_{r=0}^{\infty} \left(\frac{B_{2}^{2}}{2\mu} - \frac{\varepsilon E_{4}^{2}}{2} - \frac{B_{r}^{2}}{2\mu} \right) r dr$$
$$F_{z} = \frac{\pi}{\mu} \int_{0}^{\infty} (B_{z}^{2} - B_{r}^{2}) r dr - \pi \varepsilon \int_{0}^{\infty} E_{4}^{2} r dr \qquad (23)$$

Our calculations of the fields E_{ϕ} , B, and B, in the prototype experimental EIDI configuration described in Chapter 5 showed that the second integral in (23) is totally insignificant compared to the first integral. Some feeling for why this is true can be obtained by comparing the constants that multiply each of the integrals in (23).

$$\frac{\pi}{\mu} = \frac{\pi}{4\pi \times 10^{-7}} = 2.5 \times 10^6$$
$$\pi \epsilon \doteq \frac{\pi}{36\pi} = 2.8 \times 10^{-11}$$

Accordingly, we approximate (23) by

$$F_{z}(t) = \frac{\pi}{\mu} \int_{0}^{\infty} \left[B_{z}^{2}(t) - B_{r}^{2}(t) \right] dr \qquad (24)$$

Equation (24) is the result that we use to calculate force versus time.

4.6 IMPULSE DELIVERED TO TARGET

The impulse delivered to the stationary target is by definition given by the integral

$$\Gamma = \int_{0}^{\infty} F_{\pm}(t) dt \qquad (25)$$

where $F_{r}(t)$ is calculated using (24).

CHAPTER FIVE

A SPECIFIC SYSTEM EXAMPLE, INCLUDING EXPERIMENTAL RESULTS

5.1 DEFINITION OF THE SYSTEM

The prototype EIDI system constructed at The Wichita State University, and the results of the tests made on that system, have been described in detail by Dr. Robert Schrag [3]. Most of the material in this Chapter has been excerpted from Dr. Schrag's paper.

Figure 5A shows the prototype EIDI energy discharge system, omitting the capacitor charging circuit and the thyristor firing circuit. Two identical pulsing coils were operated in series, because that was the arrangement used in most of the de-icing tests. However, only one of the two coils was utilized in the coil-target assembly, which is detailed in Figure 5B.

The effective gap between the coil (copper) surface and the near surface of the target was .078 inch. A .032" thick 2024 T3 Aluminum disc was used as the target. Diameter of this disc was 5 inches. Two .05" thick phenolic spacer plates were used, one to maintain a fixed distance between the coil and the target, and the other to maintain the distance between the target and the rigid wooden support that prevented motion of the target. These plates could be removed, and a special magnetic field measuring plate (described in Section 5.4) inserted in their place in order to make measurements of the magnetic induction close to the surface of the target. Each coil consisted of



;

30 turns of .024" X .188" rectangular copper wire spirally wound in a single layer from an inner radius of .125" to an outer radius of 1". The initial capacitor voltage utilized for the experimental study was 400 volts.

5.2 COIL IMPEDANCE MEASUREMENTS

Impedance measurements on the coil next to the target were made using an impedance bridge. The inductance determined by these measurements, for several frequencies, appears in the table below. Impedance measurements were also made on the coil when the metal target was removed. The real part of the impedance measured on the coil without the target in place was subtracted from the real part of the impedance measured on the coil with the target in place. This increase in resistance due to the target is given in the table below.

Frequency (Hertz)	Inductance (Microhenries)	Resistance Increase (Milliohms)
500	18.6	8.2
1000	17.0	23.0
2000	14.6	48.0
4000	12.6	77.0

RESISTANCE INCREASE AND INDUCTANCE - COIL AND METAL TARGET

5.3 CURRENT WAVEFORM

Initially, current in the coil was measured indirectly by measuring the voltage across the .001 ohm non-inductive resistor in Figure 5A. Difficulties with this approach prompted the purchase of a current transformer from Pearson Electronics, Inc. Using this transformer and a storage oscilloscope, we observed the current shown in





FIGURE SC COIL CURRENT

5.4 MAGNETIC FIELD MEASUREMENTS

A magnetic field measuring plate was constructed in the manner illustrated in Figure 5D. Shallow concentric grooves were cut into both sides of a .05 inch phenolic disc, with radius increments of .2",



FIGURE 5D PARTIAL ILLUSTRATION FIELD MEASURING PLATE starting at r=.2" and ending at r=2.0". Single turn loops of .006" diameter wire were then cemented into these grooves, and their twisted leads brought out to solder tabs through radial channels.

For measuring the fields on either side of the target, the measuring plate simply substituted for the corresponding phenolic spacer plate in Figure 5B. A measurement of the axial flux density was derived from the induced voltage in any two neighboring loops connected in series opposition. For the two loops illustrated in Figure 5D, for example,

 $B_{z}(t) = 1550 \frac{\int_{0}^{t} v(z) dz}{\pi (r_{2}^{2} - r_{1}^{2})}$

where B, is in teslas, r is in inches, and V is in volts. This value is the average axial flux density over the area between the two induction loops. In the further use of this result, we will assume the flux density to apply at a radius midway between the two loops.

To measure the radial component of flux density at any radius, the front and back loops at that radius are connected in series opposition, and calculations are made from

$$B_{r}(t) = 1550 \frac{\int_{0}^{t} v(z) dz}{2\pi rh}$$

where r is the radius of the two induction loops and h is their separation, both in inches.

Plots of the magnetic induction fields obtained from these tests with the target in place are shown at selected radii in Figures 5E, 5F, 5G, and 5H. All B, data showed an anomolous behavior (irregularities) at r=.4" relative to r=.6". A separate check was made, in











which the plate was reversed (interchanging the two sides). This produced a reversal of the irregularities, so the effect was probably due to an inaccuracy in the construction of the plate.

5.5 MEASUREMENT OF IMPULSE TO TARGET

The final step in the experiment was the measurement of the impulse delivered to the target when the capacitor was discharged through the coil. This was done indirectly, using a ballistic pendulum. The impulse so measured was approximately .012 lb-sec.

CHAPTER SIX

COMPUTER ANALYSIS ON THE SYSTEM EXAMPLE

6.1 INTRODUCTION

Theoretical background for the numerical computations described in this chapter were developed in Chapters 3 and 4. This chapter will concentrate on a description of the numerical methods used to implement the theoretical development contained in these earlier chapters to predict the performance of the prototype EIDI system described in Chapter 5. The sections that follow are arranged in the order of the Figure 1 Analysis Flow Diagram, beginning with Block 4 of that diagram. This is the order in which the computations were actually accomplished.

All computations were performed in FORTRAN IV on an IBM 370. Source code was compiled with the IBM furnished G level compiler.

6.2 COIL IMPEDANCE

Equation (4) of Chapter 4, repeated as equation (1) below, is the coil impedance predicted by the transmission line model. Numerical evaluation of the integral in (1) is discussed in this section.

$$Z(\omega) = 2\pi \int_{0}^{\infty} [K'(\lambda)]^{2} \stackrel{2}{\underset{i=0}{\overset{2}{\underset{i=0}{\atopi=0}{\overset{2}{\underset{i=0}{\overset{2}{\underset{i=0}{\overset{2}{\underset{i=0}{\overset{2}{\underset{i=0}{\atopi=0}{\overset{2}{\underset{i=0}{\atopi=0}{\underset{i=0}{\overset{2}{\underset{i=0}{\atopi=0}{\atopi=0}{\overset{2}{\underset{i=0}{\atopi=0}{\underset{i=0}{\atopi=0}{\atopi=0}{\underset{i=0}{\atopi=0}$$

Two common methods are in use to allow quadrature of an improper integral such as the integral above [21]. In the first method, a transformation of variables is performed prior to construction of an

algorithm for estimating the value of the integral. This transformation is chosen in such a manner that the new integral has finite limits. The second method simply replaces the infinite upper limit with a finite upper limit, selected such that the part of the integral thus ignored contributes little to the true value of the integral. Such a method can be employed only if the integrand decays sufficiently rapidly as the variable of integration increases. Since the controlling factor in the leading behavior of the asymptotic expansion, as lambda tends to infinity, of the integrand in (1) is $e^{-\lambda}$, with g a constant, the second method was chosen for use in the numerical evaluation of (1).

A large amount of high quality mathematical software is available today (22), (23). Because of the complexity and cost of writing quality mathematical algorithms, most numerical analysts suggest that complex scientific calculations be performed using algorithms written by experts (24), (25). An 8 panel adaptive Newton-Cotes algorithm entitled QUANC8, described in (25), was chosen to perform the quadrature in (1). A user of QUANC8 may select the relative and absolute error performances desired, and the program then attempts to estimate the value of the integral within the selected error criteria. One of the parameters in the subroutine QUANC8 is an output variable that contains an estimated error bound on the returned value.

Direct computer evaluation of (1) consumes a large amount of CPU time, and is consequently expensive. This is due mostly to the appearance of the factor $[K'(\lambda)]^2$ in the integrand. (Note that $K'(\lambda)$ is defined by an integral containing a Bessel function. This makes K' oscillatory, shown in Figure 6A.) It was possible to decrease this

cost considerably by calculating $K'(\lambda)$ using QUANC8 and fitting a cubic spline function to the calculated $K'(\lambda)$ for use in evaluating (1). The algorithms used for generating the spline function coeffic-



ients and for evaluating the spline function at a given argument are entitled SPLINE and SEVAL, respectively, and are described in [25].

Once a spline function approximation for $K'(\lambda)$ was available, QUANC8 was used to individually estimate the real part (the coil resistance increase due to the metal target) and the imaginary part (the coil reactance) of (1) at selected frequencies. Appendix A contains a listing of the complete program, with the subroutines referred to, for evaluating the real part of (1). A simple modification of this program allows the imaginary part of the impedance to be evaluated.

Initial estimates of the real and imaginary parts of the integral in (1) were obtained using an upper limit of 680. Following this, the programs were run again with an upper limit of 240. Little difference was seen in the results of the two calculations, leading to the conclusion that the coil resistance and inductance estimates were good. Figure 6B shows the effects of the upper limit of integration in evaluating the imaginary part of the integral in (1) for three different frequencies.



Figures 6C and 6D on the following page show the resistance increase and inductance calculated as described above. Measured values of these parameters are also shown for comparison.

6.3 CURRENT WAVEFORM

6.3.1 Introduction

As discussed in Chapter 2, three factors preclude a simple calculation of the coil current in the EIDI prototype system. These three factors are the nonlinear diode and SCR, and the presence of the metal target next to the coil.

The effects of the target on the coil were taken into account by calculating the coil resistance increase and inductance at several frequencies, as described in the previous section. These two frequency dependent parameters were then approximated with cubic spline functions, again using the subroutine SPLINE. This provided the



COIL INDUCTANCE IN PRESENCE OF METAL TARGET

46

Original page is of poor quality ability to calculate computationally inexpensive yet sufficiently accurate coil impedances for use in frequency domain circuit calculations.

6.3.2 Current Before Clamp Diode Conduction

Although the circuit is nonlinear due to the diode and SCR, it is amenable to treatment in a piecewise linear fashion. The frequency domain model of the circuit, valid between the time the SCR is initially triggered and the time the capacitor voltage becomes zero, is shown in Figure 6E. In this Figure, the EIDI system capacitor is modeled as an ideal discharged capacitor in series with a voltage



FIGURE GE FREQUENCY DOMAIN CIRCUIT MODEL CLAMP DIODE OFF

source. This voltage source has a value of zero volts for all time prior to t=0. At t=0, it instantly rises to the amplitude and polarity V. of the initial voltage on the actual capacitor, modeling the triggering of the SCR into instantaneous full forward conduction. Voltages and currents calculated from this model are good approximations to their corresponding quantities in the physical EIDI prototype circuit as long as the diode in parallel with the capacitor is not conducting. The frequency dependent resistor appearing in Figure 6E is a composite lumped model of several loss mechanisms in the physical circuit. It includes the loss in the coil due to the presence of the metal target next to the coil (calculated as described in the previous section), resistance of the ribbon conductor used to wind the coil (corrected for skin effect), and the resistance of the cable used to connect the coil and capacitor. Resistance in the cable was modeled as frequency independent, with a measured D.C. value of .054 ohms. Coil winding resistance was calculated from

$$R_{AC} = R_{DC} \quad \text{Real} \left\{ \frac{h}{a} \quad \frac{1+j}{s} \quad \text{coth} \left(\frac{h}{a} \quad \frac{1+j}{s} \right) \right\}$$

where

 R_{oc} = D.C. coil resistance = .0235 chms h = coil thickness = .00477 meters S = copper skin depth = $1/\sqrt{\pi + \mu \sigma}$ f = frequency in Hertz \mathcal{E} = copper permeability = 4 $\gamma \propto 10^{-7}$ henries/meter σ = copper conductivity = 3.48 $\times 10^{7}$ mhos/meter

Resistance R_{AC} applies to both the coil next to the target and the idler coil (see Figure 5A).

Similarly, the frequency dependent inductance L in Figure 6E arises from more than one source. First, there is the inductance of the coil next to the target (calculated in the section above). Second, the idler coil and cable connecting the capacitor to the coil had a combined measured inductance of 23 microhenries, which was assumed to be independent of frequency. As these two inductances are in series in the circuit, they are added together to obtain a value for L.

The leakage reactance and measuring circuit impedance reflected

into the actual circuit by the current transformer used to measure the current were felt to be too small to be significant, and were not included in the Figure 6E model.

By inspection, the Fourier transform of the current in Figure 6E is given by

$$I(\omega) = \frac{V_o \left[\pi \delta(\omega) + \frac{1}{j\omega}\right]}{\frac{1}{j\omega c} + R(\omega) + j\omega L(\omega)}$$
$$= \frac{CV_o \left[\pi j\omega \delta(\omega) + 1\right]}{1 + j\omega CR(\omega) - \omega^2 CL(\omega)}$$
$$I(\omega) = \frac{CV_o}{1 - \omega^2 CL(\omega) + j\omega CR(\omega)}$$
(2)

Note that the delta distribution multiplies its own argument, and consequently contributes nothing to the time domain current i(t). Equation (2) is the basis for the current calculation, valid until the capacitor voltage becomes zero and the clamp diode in parallel with the capacitor begins conducting. For simplicity and cost considerations, an inverse discrete Fourier transform (IDFT) [36] was chosen to approximately invert $I(\omega)$. To sufficiently minimize the effects of the IDFT approximation to the continuous inverse Fourier transform, a 1024 point transform with a sample time of 5 microseconds was cho-This makes the folding frequency 100kHz, which is considerably sen. above any significant frequencies measured in the spectrum of the current in the prototype EIDI system. The 200kHz frequency window of the IDFT is sufficiently large that "windowing" effect errors in the time domain response are not too great. (These errors are primarily manifested as Gibbs' ripples on the intitial current calculated in Section 6.3.3, described in Section 6.3.4 and shown in Figure 6G.)

Appendix B contains a listing of the program used to calculate the coil current in the manner described above.

In order to determine how long a time this calculated current approximates the actual circuit current, a second calculation was performed using the circuit model in Figure 6E. The model predicts a frequency domain voltage corresponding to the physical EIDI capacitor voltage of

$$V_{c}(\omega) = V_{o}\left[\pi \delta(\omega) + \frac{1}{j\omega}\right] - \frac{V_{o}\left[\pi \delta(\omega) + \frac{1}{j\omega}\right] \frac{1}{j\omega c}}{\frac{1}{j\omega c} + R(\omega) + j\omega L(\omega)}$$

$$V_{c}(\omega) = CV_{j} \frac{j\omega L(\omega) + R(\omega)}{(j\omega)^{2} CL(\omega) + j\omega CR(\omega) + 1}$$
(3)

The term involving the delta distribution again contributes nothing to the inverse Fourier integral yielding the time domain voltage, and is dropped from consideration to yield (3). A 1024 point 5 microsecond sample time IDFT was used to approximately calculate the inverse Fourier transform of (3). Output from this program showed that the capacitor voltage would slew negatively through zero volts at t=279 microseconds. This is the time at which the diode across the capacitor is modeled as coming into conduction and acting as a short circuit. Beyond this time the model of Figure 6E is no longer valid, and consequently the current predicted by the model is no longer correct.

6.3.3 Current After Clamp Diode Conduction

Once the diode comes into conduction, the model of the EIDI circuit for all future time is as shown in Figure 6F. The inductance



FIGURE 6F FREQUENCY DOMAIN CIRCUIT MODEL CLAMP DIODE ON

and resistance in this circuit have the same physical significance that they had in Figure 6E, and their values are calculated for any desired frequency using the same algorithms. For computational convenience, time is "reset" to zero in this circuit, even though the circuit does not come into existance until t=279 microseconds in the Figure 6E circuit. Since the current in the coil is not initially zero, the circuit's magnetic energy storage is modeled by including a DC current source in parallel with a frequency dependent inductance. This current source is zero for all time prior to t=0, and for all future time has a D.C. value I. equal to the value of the current in the Figure 6E circuit at the time the capacitor voltage became zero.

Current in the Figure 6F circuit is given in the frequency domain by

$$I(\omega) = I_o \left[\pi \delta(\omega) + \frac{1}{j\omega} \right] \frac{j\omega L(\omega)}{j\omega L(\omega) + R(\omega)}$$
$$I(\omega) = I_o \frac{L(\omega)}{j\omega L(\omega) + R(\omega)}$$
(4)

Note that the term containing the delta distribution once again contributes nothing to the inverse Fourier transform of $I(\omega)$, and is dropped in (4). An IDFT was used to calculate the approximate inverse Fourier transform of $I(\omega)$ given in (4), in exactly the same manner that the expression for $I(\omega)$ in (3) was inverted. The program is very similar to the one used in the Section 6.3.2 current calculation, contained in Appendix B.

6.3.4 Combining Pre- and Post Clamp Diode Conduction Current

Because of the presence of the current source in parallel with the inductor in Figure 6F, the current in this circuit model is discontinuous at t=0. Consequently, the current cannot be bandlimited. An inherent assumption in the use of the IDFT to approximately calculate a continuous inverse Fourier transform is that the signal is bandlimited. The time domain current calculated by the IDFT showed a small amount of ripple during the first 100 microseconds due to the current spectrum not being bandlimited. Prior to "joining" in time the current predicted by the Figure 6E model with the current predicted by the Figure 6F model, this ripple was eliminated by graphically choosing current values lying close to the IDFT predicted values that joined smoothly with the current from the Figure 6E model. Figure 6G illustrates this procedure. Current ripple amplitude was 54 amps peak to peak at t=10 microseconds, decaying to 1 amp peak to peak at t=75 microseconds, with time measured in the Figure 6F model. Figure 6H shows how good the agreement is between this piecewise linear model predicted current and the current measured in the prototype EIDI circuit.

With the model predicted current now available at 5 microsecond intervals from t=0 to t=5.12 milliseconds, a 1024 point 5 microsecond sample time DFT was performed on the current samples to estimate the

Fourier transform I (ω) of the model current. This transform is needed for use in calculating the magnetic induction fields from the Hankel space transmission line coil-target model, as discussed in the next section.



FIGURE 6G PRE- AND POST- CLAMP DIODE CONDUCTION CURRENT



6.4 MAGNETIC INDUCTION

6.4.1 Introduction

Knowledge of the radial and axial magnetic induction on the coil side face of the target is required for calculation of the total force on the target. Although conceptually simple, the numerical calculation of these two fields presented more computational difficulties than were encountered in the combined total of all other calculations performed. Initial attempts at calculating the magnetic induction, performed using single precision arithmetic and using professionally written quadrature routines, produced results that were incorrect by orders of magnitude.

6.4.2 Radial Magnetic Induction on the Coil Side Face of the Target

Expression (5) shows the iterated improper integral that is to be evaluated numerically to calculate the time domain radial magnetic induction B, (0, r, t). This integral is the inverse Fourier-Hankel transform of the total current at z=0 given by (11) of Chapter 4.

$$B_{r}(o, r, t) = \mu \int_{\omega=-\infty}^{\infty} e^{j\omega t} I(\omega) \times (5)$$

$$\int_{\omega=-\infty}^{\infty} \frac{|\chi'(\lambda)|e^{-\lambda g}[1+e^{-\lambda h/3}+e^{-2\lambda h/3}]J_{1}(\lambda r) \lambda d\lambda}{1+\frac{\lambda}{\sqrt{\lambda^{2}+j\omega\mu\sigma}}} \frac{1+\frac{\lambda}{\sqrt{\lambda^{2}+j\omega\mu\sigma}} \tanh(\chi_{m}d)}{\frac{\lambda}{\sqrt{\lambda^{2}+j\omega\mu\sigma}}} + \tanh(\chi_{m}d)$$

Quadrature of iterated integrals is almost always difficult [21]. One of the most common methods of evaluating such integrals, the Monte Carlo method, could not even be considered for use in (5) due to the enormity of the very expensive and random complex function evaluations required by such an approach. Furthermore, the integrand in (5) is

 $\lambda = 0$

extremely oscillatory, having both positive and negative complex values, due to the Bessel function in the inverse Hankel transform, the function $K'(\lambda)$, and the complex exponential in the inverse Fourier transform. Quadrature of such integrands is all but impossible using Monte Carlo methods due to extreme smearing [25], [26]. It is the oscillatory nature of the integrand in (5) that makes its evaluation difficult.

After much trial and error, a workable approach to quadrature of the integral in (1) was obtained, and is described in this section. No claim is made for optimality or near optimality in this approach. After many computations of the magnetic induction had been performed using this procedure, it became apparent that simplifications could be made, while still obtaining sufficient accuracy. However, these simplifications have not been tested, and the original approach to quadrature of the integral in (5) will be described.

Measurements of the radial magnetic induction near the coil side face of the target in the EIDI prototype system provide some insight into a suitable numerical procedure for evaluating the integral in (1). The Fourier transforms of these observed fields are smooth (higher order derivatives with respect to frequency are small) with most of the energy confined to relatively low frequencies. This suggested the use of an IDFT to numerically perform the inverse Fourier transformation in (1). Further impetus for use of the IDFT is given by noting that the DFT procedure for calculating the Fourier transform $I(\omega)$ of the current, described in the preceding section, yielded transform values at equally spaced frequencies. These are the appropriate frequencies for calculating a 1024 point IDFT, with a 5

*55

microsecond sample time, of the Fourier space radial magnetic induction. However, numerical evaluation of the inverse Hankel transform in (5) at the 513 frequencies needed for calculating a 1024 point IDFT yielding a real sequence is prohibitively expensive. This expense was avoided by using the previously noted fact that the expected Fourier transform of the magnetic induction is smooth, with most of the energy confined to low frequencies. Accordingly, the inverse Hankel transform in (5) was evaluated at only 77 frequencies, chosen to provide a reasonable representation of the behavior of the Fourier transform of the known radial magnetic induction. Spline function approximations were then fitted to the real and imaginary parts of the calculated inverse Hankel transforms. The IDFT routine to calculate the desired time domain magnetic induction uses these spline functions to form B, $(0, r, \omega)$ at the 1024 frequencies needed.

With a procedure for evaluating the inverse Fourier integral in (1) available, a search for a suitable method for quadrature of the inverse Hankel integral was initiated. The infinite upper limit of this integral was simply replaced with a suitably large finite upper limit, due to the controlling factor in the leading behavior of the asymptotic expansion of the integrand being $e^{-\lambda}g$. With the problem of the infinite upper limit gone, the remaining difficulties may be divided into two somewhat overlapping classes.

Selection of a suitable algorithm for performing the quadrature is essential if accurate results are to be obtained. Most of the common quadrature algorithms are incapable of dealing with oscillatory integrals without some help. Even with the best of help, examples of problems where these algorithms completely fail abound. This has

given rise to highly specialized techniques for quadrature involving oscillatory integrands [27], [28], [29], [30]. The use of one of these specialized algorithms was avoided by writing a double precision Gaussian quadrature routine [21], [31]. This routine was then used to perform the inverse Hankel integration in (5) over the range $0 < \lambda < 10$, then over the range $10 < \lambda < 20$, etc., stopping when the desired upper limit of integration had been reached. The results of these integrations, which can be interpreted as terms in a sequence that are to be summed, were then added together first using straightforward addition, and then using the Euler series summation convergence acceleration algorithm DTEUL from the NASA Lewis Research Analysis Center Software Library [34]. For each radius at which the radial magnetic induction was evaluated, and for each of the 77 Fourier frequencies, good agreement was achieved between the results of these two summation methods.

Several different finite upper limits were used to take the place of the infinite upper limit in the inverse Hankel quadrature. It was experimentally determined that upper limits greater than 1000 resulted in very little change in the calculated induction fields.

The second problem area concerned the precision of the FORTRAN implemented on the 370. Single precision floating point word length on the IBM 370 is approximately 7 decimal digits (24 bit mantissa) [25]. This is insufficient for nearly all complex scientific calculations (32), (33). Double precision floating point word length on the 370 is approximately 15 decimal digits (56 bit mantissa) [25], and was used for all computations involving the inverse Hankel integral. Without the use of double precision arithmetic, inverse Hankel quadra-

57-

ture was impossible due to smearing.

Although the use of double precision greatly reduced the effects of smearing, it created problems of its own. A double precision Bessel function, and a double precision algorithm for $K'(\lambda)$ become necessary to evaluate (5). Simply converting a single precision algorithm for $K'(\lambda)$ to double precision does not yield sufficient accuracy to obtain good results. Double precision Bessel function algorithms were unavailable at The Wichita State University. Double precision algorithms were written for J.(x) and J.(x). The double precision algorithm that was written for $K'(\lambda)$ is described in Appendix C. Finally, the complete program for calculating the time domain radial magnetic induction, given by (5), may be found in Appendix D. Figure 6I provides a comparison between the measured radial magnetic induction close to the coil side face of the target, and the predicted radial magnetic induction on the coil side face of the target (output from the program in Appendix D).

6.4.3 Axial Magnetic Induction on the Coil Side Face of the Target

The magnetic induction B, on the target next to the coil is calculated by performing an inverse Fourier transform on (22) in Section 4.4 of Chapter 4. Then

$$B_{\pm}(o, r, \pm) = \mu \mathfrak{P}^{-1} \left\{ H_{\pm}(o, r, \pm) \right\}$$
$$= \mu \mathfrak{P}^{-1} \left\{ \frac{-1}{j\omega} \int_{0}^{\infty} V(o, \lambda, \omega) \lambda^{2} J_{0}(\lambda r) d\lambda \right\}$$

where $V(o, \lambda, \omega)$ is given by (16) in Section 4.3.1 of Chapter 4. Substituting (16) from Chapter 4 into the above,



EIGURE 61 Br VS. TIME, NEAR SIDE

$$B_{z}(o,r,t) = -\mu \int_{\omega=-\infty}^{\infty} e^{j\omega t} I(\omega) \times \qquad (6)$$

 ∞

$$\int \frac{\kappa'(\lambda)\mu\lambda e^{-\lambda g} (1+e^{-\lambda h/3}+e^{-2\lambda h/3}) J_0(\lambda r) d\lambda}{1+\frac{\sqrt{\lambda^2+j\omega\mu\sigma}}{\lambda} + \frac{\lambda}{\sqrt{\lambda^2+j\omega\mu\sigma}} + \frac{\lambda}{\tan h(\chi_m d)}}$$

$$\lambda = 0$$

$$\lambda = 0$$

$$I + \frac{\lambda}{\sqrt{\chi^2+j\omega\mu\sigma}} - \frac{\lambda}{\sqrt{\chi^2+j\omega\mu\sigma}} + \frac{\lambda}{\tan h(\chi_m d)}$$

The similarity between (5) and (6) is evident, even though the order of the inverse Hankel transforms is different. Because of this similarity, the algorithms developed to perform the quadrature in (5) were used, with only evident minor changes necessary, to perform the quadrature in (6). A comparison at several different radii between the measured axial magnetic induction close to the coil side face of the target and the calculated axial magnetic induction on the surface of the target closest to the coil is shown in Figure 6J.

6.4.4 Magnetic Induction on the Opposite Face of the Target

Although it was not needed in the calculation of the force between the coil and target, the magnetic induction on the side of the target away from the coil was calculated for comparison with the measured induction close to the same target surface. Section 4.3 in Chapter 4 derived expressions for the total voltage V, and total current I, at the junction between the metal line and air line opposite the current sources. These quantities correspond respectively to the azimuthal electric intensity and the radial magnetic induction on the surface of the target opposite the coil.



FIGURE 6J B_z VS. TIME, NEAR SIDE

Inverse Fourier-Hankel transformation similar to that in (5), but with the integrand corresponding to the current I, given in Section 4.3, yields the radial magnetic induction. Only minor changes are necessary in the program whose listing appears in Appendix D to calculate the induction. Figure 6K provides a comparison between the measured and calculated radial magnetic induction close to and on the target surface away from the coil.

Inverse Fourier-Hankel transformation similar to that in (6), but with the integrand corresponding to the voltage V, given in Section 4.3, yields the axial magnetic induction. Figure 6L shows the comparison between the measured and calculated axial induction close to and on the target surface away from the coil.

6.5 FORCE VERSUS TIME

It was shown in Section 4.5 of Chapter 4 that the total force between the coil and target is given approximately by

$$F_{2}(t) = \frac{\pi}{\mu} \int_{0}^{\infty} \left[B_{2}^{2}(o,r,t) - B_{r}^{2}(o,r,t) \right] r dr$$
(7)

For use in this integral, radial magnetic induction was calculated at 5 microsecond intervals at radii .01 inch, .2 inch, and in .2 inch increments up to a maximum of 2.0 inch, on the surface of the target closest to the coil using the algorithms described in Section 6.4.2. Axial magnetic induction was also calculated at 5 microsecond intervals at radii .01 inch, .1 inch, and in .2 inch increments up to 1.9 inch, on the same target surface using the algorithms described in Section 6.4.3. Cubic spline functions were then fitted to the squares of the radial variation of these two magnetic fields at times 0, 50,



EIGURE <u>6K</u> B_r VS. TINE, FAR SIDE



ORIGINAL PAGE IS OF POOR QUALITY

EIGURE 6L B_z VS. TIME, FAR SIDE

100, 150, ..., 900, and 1125, 1400, and 2000 microseconds for use in performing the quadrature indicated in (7), using an upper limit of 1.9 inches. The quadrature was done exactly (to the limit of the machine arithmetic) on the spline function approximations to the squared fields by integrating the cubic polynomial form of the spline function between knots, and using the spline function coefficients in the result. Appendix E contains a listing of the FORTRAN program for performing this force calculation at t=50 microseconds.

Figure 6M shows the force versus time calculated using this procedure.

6.6 IMPULSE

Equation (25) in Chapter 4, repeated below, was used to calculate

 $\Gamma = \int_{-\infty}^{\infty} F_{z}(t) dt$

the impulse delivered to the target by the coil. A spline function was fitted, with time as the variable, to the force calculated in Section 6.5 above. An exact integration was performed on the cubic polynomials of the spline function between knots, with the result that the impulse calculated was .008 lb-sec. This is lower than the .012 lb-sec impulse measured using a ballistic pendulum. It should be noted that the quadrature of the force integral in (7) above used an upper limit of r=1.9 inches instead of infinity. Although the induction decays as the radius tends to infinity, an unknown part of the force has been ignored by not taking the induction fields at radii greater than 1.9 inches into account.



FIGURE 644 FORCE BETT: EEN COIL AND TARGET

CHAPTER SEVEN

CONCLUSION AND RECOMMENDATIONS FOR FURTHER WORK

7.1 CONCLUSION

7.1.1 Summary

A method of modeling the electrical system aspects of a coil and metal target configuration resembling a practical Electro-Impulse De-Icing installation, and a simple circuit for providing energy to the coil, was presented. The model was developed in sufficient theoretical detail to allow the generation of computer algorithms for the current in the coil, the magnetic induction on both surfaces of the target, the force between the coil and target, and the impulse delivered to the target. These algorithms were applied to a specific prototype EIDI test system for which the current, magnetic fields near the target surfaces, and impulse had previously been measured.

Coil impedance was the first quantity calculated using the algorithms. Agreement between the impedance calculated and the impedance measured was seen to be very good for the resistive part, and reasonable for the reactive part. Despite the simple model used for the circuit providing energy to the coil, excellent agreement was obtained between the predicted and measured coil current.

Measured and predicted magnetic induction fields were not directly compared, due to the fields having been measured close to, but not on, the target surfaces using spatial averaging methods. The only

fields calculated were on the target surfaces, for use in calculation of the force between the coil and target. Nevertheless, there was seen to be very reasonable agreement between measured and calculated magnetic fields. The character of the time variation of these fields changed considerably with radial distance from the axis, and the algorithms correctly predicted these changes.

Calculation of the impulse easily provided the greatest disagreement between a predicted and measured quantity, with a -33% error in the calculated impulse. Impulse was measured using a ballistic pendulum containing the metal target, so, for this measurement, the metal target was no longer held rigid when the capacitor was discharged through the coil. This does not satisfy the model assumption of a stationary system. However, the period of the pendulum was sufficiently long that, during the time interval when most of the force was developed on the target, negligible motion of the pendulum should theoretically have occured. Motion of the target during the measurement of the impulse is not felt to be a satisfactory explanation for the discrepancy between the measured and calculated impulse. It was mentioned in Section 6.6, where the calculation of the impulse was described, that the infinite upper limit in the integral yielding the theoretical impulse had been replaced with a finite upper limit for the purpose of quadrature, thus incurring an unknown error. Our procedure for quadrature of the impulse integral is felt to be the most likely source of error in the calculated impulse.

7.1.2 Contributions by the Author

Reference [4] provided most of the basic methods used in this dissertation in modeling the interaction between the coil and target.
It was shown that an error had occured in the definition given in (4) of the characteristic impedance of a Hankel space transmission line, and the corrected definition was used in the theoretical development of the model. For many transmission line calculations, this error does not become evident because impedances occur in ratios (e.g., the voltage and current reflection coefficient calculations in Section 4.3.1).

Reference [4] provided no method of calculating the axial magnetic induction B.(z,r,t) from the transmission line model. An integral solution for this field, in terms of the transmission line voltage, was derived in Section 4.4. This solution allowed the use of nearly all of the algorithms developed for the calculation of the radial magnetic induction B.(z,r,t) in calculating the axial magnetic induction B.(z,r,t).

The most significant contribution of this work was the development of FORTRAN algorithms for performing the inverse Fourier-Hankel transformations yielding the induction fields, described in Section 6.4. While the author made no theoretical contributions in developing these algorithms, several diverse results from the fields of numerical analysis and approximation theory had to be applied in concert to create a working algorithm. During this development, an algorithm for the generation of Struve functions of the first and second orders was developed that, according to a computer search of the literature, is the most accurate reported.

7.2 RECOMMENDATIONS FOR FUTURE WORK

If the methods developed in this dissertation are to be applic-

able as design tools for EIDI systems, the algorithms for the calculation of the magnetic induction fields should be made more efficient. These algorithms, listed in Appendix D, take nearly half an hour of CPU time on the IBM 370 to calculate either the axial or the radial magnetic induction versus time at one specific radius. Both the radial and axial components are required, each at eleven different radii, for the force calculation described in Section 6.5. A minimum of eleven hours of CPU time is too great for the evaluation of the force-time profile of a proposed EIDI configuration. The least desirable feature of the methods presented in this dissertation is the inordinate CPU time required for the calculation of the induction fields. A sophisticated convergence acceleration routine could perhaps be devised specifically for more economic calculation of the induction fields.

Human intervention is required to proceed along the Analysis Flow Diagram of Figure 1 (page 4) in evaluating a specific EIDI system. This is because the outputs from the various programs, represented by such Figure 1 quantities as the coil impedance $Z(\omega)$, the current i(t) and I(ω), are not in the form required as the inputs for the program at the next block in Figure 1. A significant amount of design automation could be accomplished by writing one long program, using as a skeleton the programs developed for this dissertation, that will take as input the geometry of the coil-target configuration and the circuit used to provide energy to the coil, and provide as output a force-time profile and the total impulse delivered to the target.

Not all possibilities have been exhausted in the search for an analytical (or semianalytical) solution to the EIDI design problem.

Perturbation techniques have been suggested as a possible method for analytical Hankel inversion in the calculation of the magnetic induction, and should be investigated, as this is the area suffering the greatest computational expense. Furthermore, analytical solutions (if sufficiently simple) are often capable of providing insight into a problem not provided by numerical solutions.

APPENDIX A

FORTRAN SOURCE CODE FOR CALCULATING COIL RESISTANCE INCREASE DUE TO TARGET

FILE: FEB21R FORTRAN С PROGRAM TO ESTIMATE RCOIL, USING VALUES OF LAMBDA FROM O TO 680. C С VALUES OF THE E-FIELD ARE GENERATED DIRECTLY USING THE CODE С FROM THE FILE FEB2 FORTRAN. VALUES OF KPRIME ARE OBTAINED FROM A SPLINE FUNCTION APPROXIMATION. VALUES OF RCDIL CALCULATED WILL BE C С USED IN A NUMERICAL INVERSE FOURIER TRANSFORM TO CALCULATE CIRCUIT С С QUANTITIES. С С INITIALIZE С COMMEN W DATA PI/3.14159265/ TWCPI=2.*PI С MAIN PROGRAM USING QUANCE С С A=0.01 U≈680. PRINT OUTPUT HEADINGS C WRITE(6,50) A.U 50 FORMAT(' RESULTS OF NUMERICAL INTEGRATION FROM LAMBDA = ++ F4+1+ TO 1',F6.1/) GENERATE FREQUENCIES IN HERTZ С DC 10 I=1,4 DC 10 J=1,4 C GENERATE FREQUENCIES IN A 1 - 2 - 4 - 7 SEQUENCE F=10.**I IF(J.EQ.2) F=2.*F IF(J.EC.3) F=4.*F IF(J.EQ.4) F=7.*F W=TWCPI*F RELERR=1.E-8 ABSERR=0. CALL CUANCE(A,U,ABSERR,RELERR,RESULT,ERREST,NOFUN,FLAG) C INTEGRATION DONE - OUTPUT COIL RESISTANCE CALCULATED RCOIL=TWOPI*RESULT*1000. 10 WRITE(6,1010) F.RCOIL, ERREST 101G FORMAT(' FREQUENCY= ',F6.0,5X,'RCOIL= ',F7.3,' MILLIOHMS',5X, 1'EPPEST= ',E12.5) STCP END С COPIED SUBROUTINE QUANCE FOLLOWS Ċ. SUBRCUTINE QUANCE(A, B, ABSERR, RELERR, RESULT, ERREST, NOFUN, FLAG) REAL FUN, A, B, ABSERR, RELERR, RESULT, ERREST, FLAG INTEGER NOFUN REAL WO, WI, WZ, W3, W4, APFA, X0, FO, STONE, STEP, CORII, TEMP RFAL CPREV, QNOW, QDIFF, QLEFT, ESTERR, TOLERR PEAL CRIGHT(31), F(16), X(16), FSAVE(8, 30), XSAVE(8, 30) INTEGER LEVMIN, LEVMAX, LEVOUT, NOMAX, NUFIN, LEV, NIM, I, J LEVMIN = 1LEVMAX = 30LEVOUT = 6 NCMAX = 5000

NOF IN = NCMAX ~ 8+(LEVMAX-LEVOUT+2++(LEVOUT+1)) W0 = 3956.0/14175.0 W1 = 23552.0/14175.0 h2 = -3712.0/14175.0 $W_{3} = 41984.0/14175.0$ W4 = -18160.0/14175.0FLAG = 0.0RESULT = 0.0 CCR11 = 0.0ERREST = 0.0 APEA = 0.0NCFUN = 0IF(A.EC.B) RETURN LFV = 0NIM = 1 X O = AX(16) = BQPREV = 0.0FO = FUN(XO)STONE = (B-A)/16.0X(8) = (X0 + X(16))/2.0X(4) = (X0 + X(8))/2.0X(12) = (X(8) + X(16))/2.0X(2) = (X0 + X(4))/2.0X(6) = (X(4) + X(8))/2.0X(10) = (X(8) + X(12))/2.0X(14) = (X(12) + X(16))/2.0DC 25 J = 2.16.2F(J) = FUN(X(J))25 CONTINUE NOFUN = 9 $30 \times (1) = (x0 + x(2))/2.0$ F(1) = FUN(X(1))CO 35 J = 3,15,2 X(J) = (X(J-1) + X(J+1))/2.0F(J) = FUN(X(J))35 CONTINUE NOFUN = NOFUN + 8 STEP = (X(16) - X0)/16.0QLEFT = (WO*(FO + F(B)) + WI*(F(1) + F(7)) + W2*(F(7) + F(6)) $1 + W3 \neq (F(3) + F(5)) + W4 \neq F(4) \neq STEP$ QR[GHT(LEV+1)=(W0*(F(8)+F(16))+w1*(F(9)+F(15))+w2*(F(10)+F(14)) 1 +W3*(F(11)+F(13))+W4*F(12))*STEP QNOW = CLEFT + ORIGHT(LEV+1) CDIFF = QNOW - QPREV ARFA = AREA + CDIFF ESTERR = ABS(QDIFF)/1023.0 TCLERP = AMAX1(ABSERR, RELERR + ABS(AREA)) + (STEP/STONE) IF(LEV.LT.LEVMIN) GD TO 50 IF(LEV.GE.LEVMAX) GO TO 62 IF(NOFUN.GT.NOFIN) GC TC 60 IFIESTERR.LF.TOLERR) GO TO 70 50 NIM = 2+NIM LEV = LEV + 175 52 I = 1, 8 $FSAVE(I_{+}LEV) = F(I+8)$ xSAVE(I, LEV) = x(I+8)52 CONTINUE QPREV = QLEFT 00.55 I = 1, 8J = -IF(2*J+18) = F(J+9)X(2*J+18) = X(J+9)55 CONTINUE GO TO 30 60 NOFIN = 2*NOFIN LEVMAX=LEVOUT FLAG=FLAG+(B-XO)/(B-A) GO TC 70

. ,

ORIGINAL PAGE IS OF POOR QUALITY 62 FLAG=FLAG+1.0 70 RESULT=RESULT+QNOW ERREST=ERREST+ESTERR COR11=COR11+QD1FF/1023.0 72 IF(NIM.EQ.2*(NIM/2)) GD TO 75 NIM= NIM/2 LEV = LEV-1GO TC 72 75 NIM = NIM + 1 IF(LEV.LE.0) GO TO 90 **QPREV = QRIGHT(LEV)** X0 = X(16)FO = F(16)50.78.1 = 1, 8 $F(2 \neq I) = FSAVE(I, LEV)$ X(2*I) = XSAVE(1,LEV)78 CONTINUE GO TC 30 80 RESULT = RESULT + COR11 IFIERREST.EQ.0.0) RETURN 82 TEMP = ABS(RESULT) + ERREST IF(TEMP.NE.ABS(FESULT)) RETURN ERREST = 2.0*EPREST GC TC 82 END C COPIED SUBPROGRAM SEVAL FOLLOWS C REAL FUNCTION SEVAL(N+U+X+Y+B+C+D) REAL L, X(N), Y(N), B(N), C(N), D(N) DATA 1/1/ $IF(I_{\bullet}GE_{\bullet}N) = I=1$ IF(U.LT.X(I)) GOTG 10 IF(U.LE.X(I+1)) GUTC 30 10 1=1 J=N+1 20 K=(I+J)/2 IF(U.LT.X(K)) J=K IF(U.CE.X(K)) I=K IF(J.GT.I+1) GCTD 20 30 DX=U-X(I) SEVAL=Y(1)+DX*(B(1)+DX*(C(1)+DX*D(1))) RETURN ENO С C-SUBPROGRAM FUN(LAMB) TO EVALUATE THE INTEGRAND FOR QUANCE FUNCTION FUNILAMB) REAL LAMB, KPRI, MUC/12.56637E-7/, LAMBDA(48), KPRIME(48), B(4P), C(48). 1D(48) COMMON W COMPLEX CMPLX, CSQRT, CEXP, ZPDZA, IN, ZXDZA, RHOT, H, E, T1, HR2, HR0, CC, CG, 18C,8G,AC,AG,ZA,ZASQ,E2,E1,E3,H3,EP0,HR1 DATA SIGMA/1.74E7/,HD3/.00159/,G/.00278/,DD/.0008128/,IFLAG/1/ READ IN SPLINE INTERPOLATION DATA FROM FILE KPSPL COEFF C C FIRST CHECK TO SEE IF COEFFICIENTS ALREADY READ IF(IFLAG.NE.1) GOTO 10 DO 1 I=1,48 1 READ(2,10CO) LAMBDA(1), KPRIME(1), B(1), C(1), D(1) 1000 FORMAT(F6.1,4(1X,E15.6)) FIPST, EVALUATE THE PEAL AND IMAGINARY PARTS OF THE E-FIELDS REQUIRED C FOR THE INTEGRATION - USES CODE FROM FILE FEB2 FORTRAN C FOLLOW STEPS OUTLINED IN "METHOD" С IF(LAMB.EQ.O.) LAMB= 1.E-20 10 ZPDZA=(1.,0.)/CSQRT(CMPLX(1.,W*MUO*SIGMA/(LAMB*LAMB))) T=EXP(-LAMB+HD3) IN=CMPLX((T+T+T+1) \neq EXP(-LAMB+G),0.) RHOT=((1.,0.)-ZPDZA)/((1.,0.)+ZPDZA) T1=RHCT+LEXP(CMPLX(-2.+LAMB+DD,0.)/ZPDZA) ZXCZA=((1.,0.)+T1)*ZPDZA/((1.,0.)-T1) H3=IN/((1.,0.)+ZXDZA) ZA=CMOLX(0.,W*MUO/LAMB) E3=H3+ZXDZA+ZA

C CALCULATE EP2 $T = EXP(LAMB \neq G)$ AG=CMPLX((T+1./T)/2.,0.) BG=CMPLX((T-1./T)/2..0.)*ZA ZASQ=ZA*ZA CG=BG/ZASG E2=AG*E3+BG*H3 HR2=CG*E3+ÅG*H3 EP2R=REAL(E2) EP21=AIMAG(É2) C CALCULATE EP1 T=EXP(LAMB*HD3) AC=C#PLX { (T+1./T)/2.,0.) BC=CMPLX((T-1./T)/2.,0.)*ZA CC=BC/ZASQ E1=AC *E2+BC*(HR2-(1.,0.)) HR1=CC+E2+AC+(HR2-(1.,0.)) EP1P=REAL(E1) EPII=AIMAG(E1) C CALCULATE EPO EPO=AC+E1+BC+(HP1-(1.,0.)) EPOR=FEAL (EPO)

- EPOI=AIMAG(EPO) C EVALUATE KPRI USING THE SPLINE INTERPOLATION KPRI=SEVAL(48,LAME,LAMBDA,KPRIME,B,C,D)
- C NOW EVALUATE THE INTEGRAND AND PETURN FUN=LAMB*KPRI*KPRI*(EP2R+EP1R+EPOR) IFLAG=2 Peturn END

RESULTS OF NUMERICAL INTEGRATION FROM LAMBDA = 0.0 TO 680.0

					,	
FREQUENCY=	10.	RÇQIL=	0.004	MILLIOHMS	ERREST=	0.96085F+13
FREQUENCY=	20.	PCOIL=	0.017	MILLIOHMS	ERREST=	0.11895E-11
FREQUENCY=	40.	RCOIL=	0.068	MILLIOHMS	EPPE ST=	0.148125-11
FREQUENCY=	70.	RCOIL=	0.207	MILLIOHMS	ERREST=	0.16755E-10
FREQUENCY=	100.	RCOIL=	0.418	MILLIOHMS	ERREST=	0.23993E-10
FREQUENCY=	200.	RCOIL=	1.589	MILLIOHMS	ERREST=	0.30565E-09
FREQUENCY=	400.	RCOIL=	5.592	MILLIOHMS	ERREST=	0.386826-09
FREQUENCY=	700.	RCDIL=	13.841	MILLIOHNS	ERREST=	0.34003E-09
FREQUENCY=	1000.	8 CO 1 L =	22.783	MILLIOHMS	ERR E ST≠	0.33792E-09
FREQUENCY=	2000.	RCOIL=	47.373	MILLIOHMS	ERREST=	0.52783F-08
FREQUENCY=	4000.	RCOIL=	70.588	MILLIOHMS	ERREST=	0.41543E-C8
FREQUENCY=	7000.	FCOIL =	81.442	MILLIOHMS	ERR EST =	0.58640E-08
FREQUENCY=	10000.	RCOIL=	85.525	MILLIOHMS	ERREST≈	0.64146E-08
FREQUENCY=	20000.	RCOIL=	92.622	MILLIOHMS	ERR EST=	0.39609E-08
FREQUENCY=	40000.	RCOIL=	109.270	MILLIOHMS	ERR EST =	0.49584E-08
FP EQUENCY=	70000.	PCOIL=	142.962	MILLIOHMS	EPPEST=	0.649458-08

ORIGINAL PAGE IS OF POOR QUALITY

APPENDIX B

FORTRAN SOURCE CODE FOR CALCULATING COIL CURRENT BEFORE CLAMP DIGDE CONDUCTION

	PROGRAM TO PERFORM NUMERICAL FOURIER INTEGRAL INVERSION TO CALCULATE THE CURRENT IN THE COIL IN THE WING CIRCUIT (BEFORE THE DIDDE CLAMP CUTS IN). THIS PROGRAM INCLUDES THE FREQUENCY DEPENDENCE OF THE COIL RESISTANCE AND INDUCTANCE. OUTPUT FROM THIS PROGRAM WILL BE USED TO CALCULATE THE SPECTRUM OF THE CURRENT. THIS PROGRAM IS ADAPTED FROM SERCUR FORTRAN.
v	COMPLEX F1(2048),CMPLX,CONJ3 RFAL L,LSIMID DATA N/2048/,T/5.F-6/,C/600.E-6/,RCA3LE/.054/,RDC/.047/,LSIMID/23. *E-6/,PI/3.141593/ TTOT=FLOAT(N)*T DELTAF=1./TTOT DELTAW=2.*PI*DELTAF KDDE=-1
ç	FORM COMPLEX SPECTRUM HATRIX
נ <u>כ</u>	IEND=N/2+1 DC 5 I=1,IEND w=FLOAT(I-1)*DELTAW CHECK FOR FREQUENCY w=0 IF(w.NE.0.) GOTC 3 R=QCABLE+RDC+.004E-3 L=LSIMID+17.865E-6
C C	W .NF. O - CALCULATE RESISTANCE R AS THE SUM OF 3 TERMS: A CONSTANT D.C. TERM (RCABLE); SKIN EFFECT IN COIL (RAC); AND LOSS IN SKIN OF
C C	WING (COILR) 3 R=PCABLE+RAC(W)+CDILP(W) CALCULATE THOUSTANIE TO SUM DE 2 TERMET A CONSTANT TERM DUE TO
CCC	THE SIMMONDS AND IDLER (LSIMID); AND A TERM DUE TO THE COIL IN THE WING (COILL)
ن د د	LELSIMID+COILL(W) NOW FORM THE CONSTANTS USED IN EVALUATION OF THE FOURIER TRANSFORM OF THE TIME VARYING PART OF THE CIRCUIT CURRENT 4 Pl=1+C
C	P2=R≠C NOW FORM THE MATRIX VALUES 5 F1(1)=(1,,0,)/CMPLX(1,-₩*₩*P1,₩*P2) ISTART=T./2+2 DG 7 I=ISTAFT,N 7 F1(1)=CUNJG(F1(N+2-1))
C	CALL FFT(KUDE,N,DELTAF,F1,10)
Č	EXPPESS TICT IN MILLISEC, T IN MICROSEC
c	TP=T*1.66 BEGIN OUTPUT OF RESULTS WRITE(6,201) N,TTOTP,TP 201 FURMAT('INUMBER OF PDINTS=',I4/' TOTAL TIME INTERVAL= ',F7.3,' MIL 1LISEC',/' SAMPLING PERIOD= ',F7.3,' MIGROSEC'//' OUTPUT FROM PROGR
	2AM MAY22A FORTPAN //) WRITE(6,100)
	100 FORMAT(' CIRCUIT CURRENT - FREQUENCY DEPENDENT L, R'// T 11ME(MSEC) + 5X, TOTAL CURRENT RESPONSE'/, T23, 'REAL', 15X, 'IMAG'/)
	<pre>F3C=400*C CO 20 I=1,100 DT=FLCAT(I~1)*T#1000. F1(I)=CMPLX(FSC,0.)#F1(I) 20 wRITE(6,200) DT,F1(I) 20 GCRMAT(1X,F12.3,1X,2(F15.5,5X)) 10 STCP END</pre>

	FUNCTION RAC(W)	· · · · · · · · · · · · · · · · · · ·
	C SUBPROGRAM TO EVALUATE SKIN EFFECT RESI C FUNCTION OF THE RADIAN FREQUENCY #	STANCE RAC OF COIL AS A
C	COMPLEX CMPLX, CEXP, TAU, TAUHD2, T1, T1R	
	DATA H/.00477/.SIGMA/3.48E7/.RDC/.04	7/
	TAU=(1.,1.)/CMPLX(DELTA.0.) TAU=D=TAU=CMPLX(DELTA.0.)	
	T 1=CE XP(TAUHD2)	
	RAC = RDL + REAL (TAUHD2 + (T1+T1R) / (T1-T1R	
	C SUBPROGRAM TO EVALUATE COIL RESISTANCE VING SKIN) AS A FUNCTION OF THE RADIAN	COILR (DJE TO PROXIMITY OF FREQUENCY W
ι	REAL FREQ(24), RCOIL(24), BR(24), CR(24),)R(24)
	IF (IFLAG.NE.1) GGTD 5	
č	C INTERPOLATION	EFF FUR SPEINE FUNCTION
	1 READ(2,100) FREQ(1), RCOIL(1), BR(1),C	R(I), DR(I)
	TWOPI=2.*PI	
	5 F=W/IWOPI	
	CCILR=.004E-3	
	10 COIL = SEVAL(24,F,FREQ,PCOIL,BR,CR,DR) * 1. E-3
	ENC ENC EFAL FUNCTION SEVAL(N.11.X.Y.B.C.D.)	
ĉ	C COPIED SUBPROGRAM SEVAL FOLLOWS - EVALU	ATES SPLINE
Ċ	C REAL U.X.(N) • Y(N) • B(N) • C(N) • D(N)	
	DATA I/1/	· · · · ·
		Ň
	$\frac{J = N + 1}{20 \text{ K} = (1 + J)/2}$	· · · ·
	IF(U, GE, X(K)) I = K	
	$\begin{array}{c} 1 \\ 30 \\ \text{DX=U-X(I)} \\ \text{SYM} = \frac{1}{2} \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ $	
r	FUNCTION COILL(W)	
	C SUBPROGRAM TO EVALUATE COIL INDUCTANCE C RADIAN FREQUENCY W	COILL AS A FUNCTION OF THE
-	PEAL FREQ(16), LCDIL(16), BL(16), CL(16 DATA PI/3, 141593/, IFLAG/1/), DL(16)
с	IF (IFLAG.NE.1) GOTO 5 C PEAD IN CUEFFICIENTS FROM FILE LCOIL CO	EFF FOR SPLINE FUNCTION
C	C INTERPOLATION DO 1 I=1,16	
	1 RE40(3,100) FREQ(1),LCOIL(1),BL(1),C 100 FORMAT(F6.0,4(1X,E15.6))	L(I), DL(I)
	TWCPI=2.*PI IFLAG=-1	
	5 F=W/TWOPI IF (F.GT.10.) GPTO 10	
	COILL=17.865E-6 PETURN	
	10 IF(F.GE.70.E3) GOTO 20 COILL=SEVAL(16,F,FREQ,LCOIL,BL,CL.DL)*1.E-6
	20 COLL=9.263E-6	
	RETURN END	

	GRAM TU	MPLE	MENT	THE	FAST	FD		R T TWC	2ANS	FURM			THE		2FR 01	F 5
بر . ا	COMPLEX :	((N),	W,X1	,CMPL	х Х	INL.	ANU	31-	161	4 14 AL	. 1 . 1	- 3 3		3(7.1)		
5	IF=0 N1=N N2=N1/2															
<u>,</u>	TF(N2*2.1 TP=TR+1	NE.NI	L) SC	TC 10	0											
	N1=N2 IF(+1.GT	1)60	9TO 5											•		
	L=N/2	1827	Ň													
	IR1=IR-1 K1=0	-) 10									•					
15	DD 20 I=: K=K1+1	- 1 , 1 () L , L	(×.						
	KPL=K+L AM=KBITR	(K1/2	**IR	1, [R)				;								
	IF(AM.NE X1=X(KPL	•0•)	GUIU	18												
19	ARG=AM*PI C=COS(AR	N 51														
	S=-KODE*: W=CMPLX(SIN(/	ARG)													
19	X(KPL)=X X(KPL)=X	//////////////////////////////////////	(1							:						
20	K1=K1+1 K1=K1+L															
	IF(K1.LT K1=0 TP1=TP1-	• N) (1	50TO -	15			*									
30	L=L/2 D0 40 K=	L 1 • N														
	K1=KBITR IF(K1.LE	(K—1 •K) (IR)+ 50TC	1 40												
	X(K) = X(K) X(K) = X(K)	1)														
40	CONTINUE IF(DELTA	.EQ.1	L.) R	ETURN	1			:								
50	DO 50 K= X(K)=DEL	1,N TA*X	(K)													
100 101	WRITE(6, FORMAT(/	101)	N ⊭≠≠ N	= •,	[6, "	1 S	NOT	A	PGWE	R DF	2,	FFT	FUN	AB	ORTED	**
1	RETURN	•									•	•				
	FUNCTION	KBI	TF(K,	IR)												
	KBITR=0 Kl=K															
	DO 1 I=1 K2=K1/2 KBITP=2#	, IP KRTTI	D + K 1 -	7287						-						
1	K1=K2 RETURN		· · · I ··	E * ^ 2												
	END															
										•						
					•											

APPENDIX C

CALCULATION OF $K'(\lambda)$

The function we have called $K'(\lambda)$ arises from the transformation of real space current discs into Hankel space. By definition,

$$K'(\lambda) = \frac{30}{3(R_2 - R_1)} \int_{R_1}^{R_2} x J_1(\lambda x) dx$$
 (1)

Straightforward quadrature of (1) using the Newton-Cotes algorithm QUANC8 was initially performed for generating values of K'(λ) for use in approximating K'(λ) with a cubic spline function. This provided sufficient accuracy for use in numerical calculation of coil impedance, as described in Section 6.2. Attempts at calculating the radial magnetic induction using this spline function approximation for K'(λ) were a total failure. This appendix describes a procedure for calculating the function K'(λ) that is accurate to 14 digits when implemented in double precision FORTRAN on an IBM 370.

The integral in (1) can be evaluated in closed form in terms of named special functions. Change variables,

 $\lambda x = y \Rightarrow dx = dy/\lambda$ $x = R_1 \Rightarrow y = \lambda R_1$ $x = R_2 \Rightarrow y = \lambda R_2$

so that (1) becomes

$$K'(\lambda) = \frac{10}{R_2 - R_1} \int_{\lambda R_1}^{\lambda R_2} \frac{y}{\lambda} J_1(y) \frac{dy}{\lambda}$$

$$K'(\lambda) = \frac{10}{R_2 - R_1} \frac{1}{\lambda^2} \int_{\lambda R_1}^{\lambda R_2} y J_1(y) dy \qquad (2)$$

From reference [18],

$$\int J_{1}(y) y \, dy = \frac{\pi y}{2} \left[J_{1}(y) H_{0}(y) - H_{1}(y) J_{0}(y) \right]$$

where $H_0(x)$ and $H_1(x)$ are Struve functions of orders 0 and 1 respectively [18], [19], and $J_0(x)$ and $J_1(x)$ are Bessel functions of the first kind of orders 0 and 1 respectively. Using this result in (2) yields

$$K'(\lambda) = \frac{5\pi y}{\lambda^2 (R_2 - R_1)} \left[J_1(y) H_0(y) - H_1(y) J_0(y) \right] \Big|_{y=\lambda R_1}^{\lambda R_2}$$
(3)

In order to use this result, double precision algorithms for generating the Bessel and Struve functions must be available. Double precision Bessel functions are readily available, but Struve functions are not. A computer search of the literature resulted in a reference to a Naval Research Laboratory report that contained FORTRAN source code for generating integer order Struve functions with positive arguments [36]. We were unable to make use of this code because it used subroutines to which we did not have access.

For several years, mathematicians have been aware of the desirability of using truncated Jacobi series of Chebyshev polynomials for numerical approximation of various special functions (37), (38). Luke (37) provides coefficients b, and c, for Chebyshev series expansions

$$H_{n}(x) = \frac{x}{8} \sum_{n=0}^{\infty} b_{n} T_{2n} \left(\frac{x}{8}\right), \quad |x| \leq 8$$

$$H_{n}(x) = \sum_{n=0}^{\infty} c_{n} T_{2n} \left(\frac{x}{8}\right), \quad |x| \leq 8$$

where $T_{2n}(x)$ is a Chebyshev polynomial of the first kind of order 2n. All coefficients having magnitudes greater than 10^{-20} are given, allowing generation of $H_0(x)$ and $H_1(x)$ with 20 decimal digit accuracy for arguments whose magnitude is less than 8 [39]. Using the identity

 $T_{e_n}(x) = T_n(2x^2-1)$

and retaining sufficient terms for 15 digit accuracy, these series expansions become

$$H_{o}(x) \stackrel{:}{=} \frac{x}{8} \sum_{n=0}^{14} b_{n} T_{n} \left[2 \left(\frac{x}{8} \right)^{2} - 1 \right] , \quad |x| \leq 8 \quad (4)$$

$$H_{1}(x) \stackrel{:}{=} \sum_{n=0}^{14} c_{n} T_{n} \left[2 \left(\frac{x}{8} \right)^{2} - 1 \right] , \quad |x| \leq 8 \quad (5)$$

Then for arguments λ such that $\lambda R_i < 8$, the expression on the right hand side of (3) is evaluated using (4) and (5). Chebyshev polynomials are evaluated in double precision arithmetic using the subroutine DCNPS from the NASA Lewis Research Analysis Center Software Library [40].

Luke also lists coefficients d, and e, for the series

$$H_{0}(x) - Y_{0}(x) = \frac{2}{\pi x} \sum_{n=0}^{19} d_{n} T_{2n} \left(\frac{8}{x}\right), \quad x \ge 8$$

$$H_{1}(x) - Y_{1}(x) = \frac{2}{\pi} \sum_{n=0}^{16} e_{n} T_{2n} \left(\frac{8}{x}\right), \quad x \ge 8$$

for 15 digit accuracy. The functions $Y_0(x)$ and $Y_1(x)$ are Bessel functions of the second kind of orders 0 and 1 respectively. Rather than use these series directly to evaluate $H_0(x)$ and $H_1(x)$ for arguments greater than 8, some simplification is possible. Writing

$$H_{o}(x) \doteq \frac{2}{\pi x} \sum_{n=0}^{19} d_{n} T_{an} \left(\frac{8}{x}\right) + Y_{o}(x) \triangleq A_{o}(x) + Y_{o}(x)$$

$$H_{1}(x) \stackrel{:}{=} \frac{2}{\pi} \sum_{n=0}^{16} e_{n} T_{2n} \left(\frac{8}{x}\right) + Y_{1}(x) \triangleq A_{1}(x) + Y_{1}(x)$$

and substituting these expressions for $H_0(x)$ and $H_1(x)$ into the expression $J_1(x)H_0(x)-H_1(x)J_0(x)$ appearing in (3), and suppressing the argument x, we have

$$J_1 H_0 - H_1 J_0 = J_1 [A_0 + Y_0] - [A_1 + Y_1] J_0$$

 $= J_{1} A_{0} - A_{1} J_{0} + J_{1} Y_{0} - Y_{1} J_{0}$ $= J_{1} A_{0} - A_{1} J_{0} + \frac{2}{\pi \times}$

using a well known property of the Wronskian of Bessel functions of the first and second kinds [18]. Thus, the expression on the right hand side of (3) can be written as

$$\frac{5\pi y}{\lambda^2 (R_2 - R_1)} \left[J_1(y) A_0(y) - A_1(y) J_0(y) + \frac{2}{\pi y} \right]$$

when it is to be evaluated at an argument $y = \lambda R_n > 8$.

Using these results, $K'(\lambda)$ can be evaluated from (3) for nonnegative values of the argument to at least 14 digit accuracy.

21

APPENDIX D

FORTEAN SOURCE CODE FOR CALCULATING RADIAL TIME DOMAIN MAGNETIC INDUCTION ON THE COIL SIDE SURFACE OF THE TARGET

C FILE: WAUG2	E FORTRAN N	/1.3
С РАЧСКАМ ТО С С АТ ТНЕ СОІС С ЕЧОМ ТНЕ СО С ЕЧОМ ТНЕ СО С ЕЧОМ NASA E	ALCULATE THE →ADIAL (SIDE FACE OF THE SK JIL AXIS• ADAPTED FH LEWIS AUGUST 85•	COMPONENT OF THE MAGNETIC INDUCTION IN, AT A PADIAL DISTANCE RINCH DM SOURCE-AUG2 VI-O BROUGHT TO WSU
L (\$6666	CAUTION COCC	¢ .
C DARRAY SHOU C DIMENSION N C SHOULD HAVE C IN ADDITION C IERFL3(2,	ULD HAVE DIMENSION NAM NINTL. IN THE FUNCTIO DIMENSION NARRAY. I N. THE ARRAY IERFLG SU 77,NARRAY).	RRAY, AND DHINTL SHOULD HAVE ONS DECT(I) AND DEUN(DX), DARRAY NARRAY+1=NINTL HOULD BE DIMENSIONED AS
ς οφοώο Γ	CAUTION \$\$\$\$\$	2
C VI+1 - ADD WIT C VI+2 - CHA C INV C CHA C VI+3 - ADD C AT	DED THE ABILITY TO ST THE EACH INDIVIDUAL GAU ANGED ABSOLUTE AND REU VERSE HANKEL TRANSFOR ANGED PRINT FORMATS T DED OUTER LOOPING ABIN SEVERAL RADII.	DRE AND PRINT THE FLAGS ASSOCIATED USSIAN QUADRATURE. LATIVE ERROR REQUESTS TO 1.D-12 IN M QUADRATURE. O ALLOW SPOOLING OUTPUT TO MY READER. LITY TO CALCULATE MAGNETIC INTENSITY
L IMPLICIT COMPLEX C REAL®3 DA REAL©4 DEL SEAL MONT	REAL¢8 (D) URR(77),CURNT,CFHR(1) RRAY(100),DHINTL(101 LT4 LT4 IM/ IAN // DAY/* 14 *	024),CMU0/(12.56637E-7,0.)/),DHR?O(77),DHIRO(77) /.VERS/141.31/
REAL DAEG BHRSD(77) INTEGER I DATA NAPR EXTERNAL COMMIN DA	GA(77),HRRO(77),HIRO(,HISB(77),HISC(77),HI IEREG(2,77,100) AAY/100/,NINTL/101/ OFUN,DFCT ARRAY,CURNT,IFUN,R,TWO	77).HRSB(77).HRSC(77). SD(77).RINCH/+01/ DPIF
BUNCK 1 -	READ CUIL CURRENT SP ONLY) FROM FILE CURR FREQUENCIES ARE STOR CURRENT SPECTRAL VAL	ECTRUM (STEP T) (POSITIVE FREQUENCIES SPEC DATA (LOGICAL DEVICE S). ED IN OMEGA(I) WITH THE CORRESPONDING UE IN CURR(I).
READ CURRENT	T SPECTRUM (HEADINGS)	HAVE BEEN REMOVED FROM THE FILE
DO 10 I=1 10 READ (5,1) 1000 FORMAT(T1	L+77 1000) ONEGA(1)+CURR(1 19+612+5+T36+613+5+T5) 1,613.5)
OUTER LOOP TO CALCULATED	O CONTROL RADIUS AT	WHICH MAGNETIC INDUCTION IS
00 59 194	005=16+20+2	
C PINCH=FLUA C EXPRESS R IN P=RINCHP+()4T(THADUS)/10. 1 METERS 10254	
С 9L1CK 2 - (CALCULATE INVERSE HAD INTENSITY. THIS IS FREQUENCIES OMEGA(I) INMEGA IS PASSED TO TO VAPIABLE TWOPIE. THI SPECTRUM AT THIS FREE COMMON VAPIABLE CURNI	UKEL TRANSFORM OF THE MAGNETIC DONE AT EACH OF THE DIFFERENT I I=1+2++++,77+ THE FREQUENCY HE SUBPROGRAM FUN THRU THE COMMON E COMPLEX VALUE OF THE CURRENT DUENCY IS PASSED TO FUN THRU THE I+
CALCULATE THI USED IN FURS DSTEP=10+1 DD 1 I=1+1 1 DHINTL(I)	HE REAL¢A ARRAY DE UPI UNING THE SEQUENCE DE DO NINTL DELOAT(I-1)¢DSTEP	PER AND LOWER INTEGRATION LIMITS Magnetic intensities.

PRINT HEADINGS r PRINT HEADINGS WRITE(5,90D0) VERS,MONTH,DAY,DHINTL(NINTL) 9000 FORMAT(* 10UTPUT FROM WAUG2 FORTHAN *,A4,10X,44,44,/,* UPPER INVE 0455 HANKEL TRANSFORM INTEGRATION LIMIT = *,D13+5//) WRITL(5,90D1) RINCH,PINCH 9001 FORMAT(* FREQUENCY REAL PART OF MAGNETIC INTENSITY*,T57,*IMAG PA 08T OF MAGNETIC INTENSITY*,/,* (HERT2)*,T20,*M(R=*,F3+1,*,F)*,T38 0;IER N*,T53,*M(R=*,F3+1,*,F)*,T81,*IFR N*) BEGINNING OF MAIN LOOP TO CALCULATE INVERSE HANKEL TRANSFORM AT 77 DIFFERENT FREQUENCIES D0 900 IMAIN=1,77 TWOPIF=OMEGA(IMAIN) CURNT=CURR(IMAIN) C CURNTECURR(IMAIN) CALQULATE HEAL PART OF HR(R,TWOPIF) C CALLULATE MEAL FART OF THE ARTIAL INVERSE HANKEL TRANSFORMS BY INTEGRATING IFUN=1 FORM THE SEQUENCE OF PARTIAL INVERSE HANKEL TRANSFORMS BY INTEGRATING BETWEEN LAMBDA SUB(I) AND LAMBDA SUB(I+1), FOR I=1 TO NARRAY (THESE LIMITS ARE IN THE ARRAY DHINTL). SET UP ABSOLUTE AND RELATIVE PUADRATURE ERROR PEQUESTS C ē DAUSTHEIL-D-IZ DAUSTHEIL-D-IZ INITIALIZE THE VARIABLE USED TO ACCUMULATE THE RESULTS OF THE NUMERICAL INTEGRATION OVER EACH LAMBDA SUB-RANGE+ Ć DINIGED.DU DO 2 [=1,NARRAY FORM THE LOWER AND UPPER INTEGRATION LIMITS DA=DHINTL(I) DB=DHINTL(I+1) INTEGRATE (THESE VALUES ARE PASSED TO THE FU ASSUT TO THE FURTHER ARE PASSED TO THE FU C NTEGRATE (THESE VALUES ARE PASSED TO THE FUNCTION DECT THRU COMMON). Result of the integration is stored in Darray(i) for future use in the euler convergence acceleration algorithm dieul. c THE EULY LUNVERGENCE ACCELERATION ACOUNTHY DIEUL+ CALL JGAUSS(DA,DB,DFUN,DRER,DABSER,DARRAY(I),DERR,IER) STORE ERROP FLAG [ERFLG(1,IMAIN,I)=IFR NOW SUM THE INTEGRALS FOR COMPARISION WITH THE ACCELERATED SUM TO BE CALCULATED IN SUBROUTINE DIEUL+ С C 2 DINTREDINTREDARARY(1) NOW USE THE FULER CONVERGENCE ACCELERATION ROUTINE TO BEST ESTIMATE C THE REAL PART OF THE TOTAL INVERSE MANKEL TRANSFORM AT THE FREQUENCY TWOPIF. č CALL DTEUL(DECT, DSUMR, NARRAY, L.E-14, IERR, NA) DHRRD(IMAIN)=DSUMR HARD(IMAIN)=SNGL(DSUMP) CALCULATE IMAGINARY PAPT OF HA(R, TWOPIF) C IFUNEZ IFUNEZ FORM THE SEQUENCE OF PARTIAL INVERSE HANKEL TRANSFORMS BY INTEGRATING RETWEEN LAMBDA SUB(I) AND LAMBDA SUB(I+1), FOR I=1 TO NARRAY (THESE LIMITS ARE IN THE ARRAY DHINTL). INITIALIZE THE VARIABLE USED TO ACCUMULATE THE RESULTS OF THE NUMERICAL INTEGRATION OVER EACH LAMBDA SUB-RANGE. ć NITIALIZE THE VARIABLE USED TO ACCUMULATE THE RESULTS OF THE NUMERICAL INTEGRATION UVEP EACH LAMBDA SUB-RANGE. DINTIED.DO DO 3 [=1,NARRAY FUH THE LOWER AND UPPER INTEGRATION LIMITS DAEDHINTL(I) DBEDHINTL(I+1) INTEGRATE (THESE VALUES ARE PASSED TO THE FUNCTION DECT THRU COMMON). RESULT OF THE INTEGRATION IS STORED IN DARPAY(I) FOR ENTURE USE IN THE EULER CONVERGENCE ACCELERATION ALGORITHM DIEUL. CALL DGAUSS(DA,DB,FEUN,DREP,DARSEP,DARRAY(I),DERR,IER) STORE ERROR FLAG IEPFLG(2,IMAIN,I)=IER NOW SUM THE INTEGRALS FOR COMPARISION WITH THE ACCELERATED SUM TO BE CALCULATED IN SUBPOUTINE DIFUL. 3 DINTIEDINTIFDARRAY(I) NOW USE THE EULER CONVERGENCE ACCELERATION ROUTINE TO BEST ESTIMATE THE REAL PART OF THE TOTAL INVERSE MANKEL TRANSFORM AT THE FREQUENCY TWOPIF. CALL DIEUL(DECT,DSUMI,NAPRAY,1.E-14,IERI,NI) DHIRO(IMAIN)=SNGL(DSUMI) FEOMEG(IMAIN)=SNGL(DSUMI) FEOMEG(IMAIN)=SNGL(DSUMI) FEOMEG(IMAIN)=SNGL(DSUMI) FEOMEG(IMAIN)=SNGL(DSUMI) FEOMEG(IMAIN)=SNGL(DSUMI) FEOMEG(IMAIN)=SNGL(DSUMI) FEOMEG(IMAIN)=SNGL(DSUMI) FEOMED(IMAIN)=SNGL(DSUMI) FEOMEG(IMAIN)=SNGL(DSUMI) FEOMEG(IMAIN) C ē C C C OINTI 0010/1 9010 F03MAT(T4,F7+0,T14,2(D23+15,2X,F1,2X,F3,12X),/+T10,*SUM=*,D23+16, 9753,*SUM=*,D23+16,/) FND OF DEVELOPMENT STAGE CODE BLUCK 3 - FORM HR(R,N¢DELTAW) NEEDED FOR IDFT. US INTEPPOLATION ON THE RESULTS OF BLOCK 2-USE SPLINE FUNCTION C NSPL=77 FURM SPLINE CUEFFICIENTS FOR THE REAL PART OF HR(R.Q) CALL SPLINE(NSPL, MMEGA, HRRD, HRSB, HRSC, HRSD) C

..

FORM SPLINE CUEFFICIENTS FOR THE IMAGINARY PART OF HR(R.O) Call Spline(NSPL, MAEGA, HIRD, HISB, HISC, HISD) Set UP PARAMETERS FOR USE IN IDET CALCULATIONS C SET UP PAJAMETERS FOR USE IN IDET CALCULATIONS TSAMP=3.6-5 NDET=1024 TWOPI=2.03.141593 WD=TWOPI/(TSAMPONDET) ISTOP=NDET/2+1 FORM THE SPECTRUM DE THE FIRST (N/2+1) FREQUENCIES DO 40 I=1.ISTOP C C PADERUEWOOFLUAT(I-1) 40 CENR(I)=CMPLX(SEVALR(NOPL, PADEPO, OMEGA, MPRO, MROB, MRSC, MPSD), SEVALI (NSPL, PADERU, OMEGA, MIPO, MISB, MISC, MISD)) NOW FORM THE CONJUGATE PART OF THE SPECTRUM ISTART=ISTOPH1 DOT FOR THE SECTOR C DO SO I=ISTART,NDFT SO CEHR(I)=CONJG(CEHR(NDET+2-I)) С CALCULATE THE INVERSE FOURIER TRANSFORM OF THE MAGNETIC Intensity. Uses an IDET to simplify the numerical Procedure. Ĉ BLUCK 4 -Ć C DELTA=1./(TSAMP©FLOAT(NDFT)) KODE=-1 CALL FFT(KODE,NDFT,DELTA,CFHP,A60) C PRINT RESULTS - TIME DDMAIN MAGNETIC INTENSITY WRITE(6,2000) 2000 FURMAT(//* TIME(MILLISEC) INDUCTION (TESLA)*) DU 70 I=1,201 TIME=.005©FLOAT(I-1) CFHR(I)=CFHR(I)©CMU0 70 WRITE(6,2010) TIME,CFH2(I) 2010 FORMAT(4X,F8.3,T22,F8.5,T41,E11.4) DU 30 I=22h,1001,5 TIME=.005©FLOAT(I-1) CFHR(I)=CFHR(I)©CMU0 40 WRITE(6,2010) TIME,CFHR(I) C PRINT 0FROM FLAGS WRITE(6,4500) 4500 FORMAT(* 1ERROR FLAGS FROM REAL INTEGRATION:*/) DO 215 I=1,NARRAY 216 WRITE(6,4501) (IERFLG(1,II,I),II=1,77) 4501 FORMAT(* 1ERROR FLAGS FROM IMAG INTEGRATION:*/) DO 217 I=1,NARRAY 217 WRITE(6,4501) (IERFLG(2,II,I),II=1,77) 4502 FORMAT(* 1ERROR FLAGS FROM IMAG INTEGRATION:*/) DO 217 I=1,NARRAY 217 WRITE(6,4501) (IERFLG(2,II,I),II=1,77) C 37 CONTINUE *0 STOP END REAL FUNCTION SEVALR(N,U,X,Y,B,C,D) C COPIED SUBPROGRAM SEVAL FOLLOWS DELTA=1./(TSAMP*FLOAT(NDFT)) C C С 10 I=1 J=Ñ+1 J=N+1 20 K=(I+J)/2 IF(U+LT+X(K)) J=K IF(U+GE+X(K)) I=K IF(U+GE+X(K)) I=K IF(J+GT+I+1) GOTO 20 30 DX=U-X(I) SEVALA=Y(I)+DX≑(B(I)+DX≑(C(I)+DX≑D(I))) RETURN END END PEAL FUNCTION SEVALI(N,U,X,Y,3,C,D) C CUPIED SUBPROGRAM SEVAL FOLLOWS REAL U,X(N),Y(N),8(N),C(N),D(N) DATA I/I/ IF(I+GE-N) I=1 IF(U+LT-X(I)) GOTO 10 IF(U+LE-X(I+1)) GOTO 30 IO IF(I=1 J=N+1 10 J=N+1 K=(I+J)/2 IF(U+LT+X(K)) J=K IF(U+GE+X(K)) I=K IF(J+GT+I+1) GUTO 20 DX=U+X(I) SEVALI=Y(I)+DX≑(B(I)+DX≑(C(I)+DX≑D(I))) REIU=N 20 30 END

SURPOUTINE SPLINE(N,X,Y,B,C,D) CUPIED SUMMUUTINE SPLINE FOLLOWS INTEGF= N AEAL X(N),Y(N),B(N),C(N),D(N) INTEGF= NM1+ IB, I PEAL T NM1=N-1 IF(N-LT-2)PETURN IF(N-LT-3)GO TO 50 D(1)=X(2)-X(1) C(2)=(Y(2)-Y(1))/D(1) D(1)= 2, NM1 U(T)= 2, NM1 U(T)= 2, NM1 U(T)= 2, (D(1-1)+D(T))/D(T) C(T)= C(T+1)-C(T) B(N)=-D(N-1) C(1)=0. С C(1)=0. C(N)=0. $C(N) = 2^{-1}$ $IF(N \in \{0\}, 3\} = G0 = T0 = 1^{-1}$ C(1) = C(3)/(X(4) - X(2)) = C(2)/(X(3) - X(1)) C(N) = C(N-1)/(X(N) - X(N-2)) = C(N-2)/(X(N-1) - X(N-3)) $C(1) = C(1) \Rightarrow 0(1) \Rightarrow 2^{-1}(X(4) - X(1))$ $C(N) = -C(N) \Rightarrow 0(N-1) \Rightarrow 2^{-1}(X(N) - X(N-3))$ $C(N) = -C(N) \Rightarrow 0(N-1) \Rightarrow 2^{-1}(X(N) - X(N-3))$ 15 = 0(1-1)/8(1-1) $R(1) = B(1) - T \Rightarrow 0(1-1)$ $C(1) = C(1) - T \Rightarrow C(1-1)$ C(N) = C(N)/8(N) $D0 = 30 = 1^{-1} + MM1$ I = N - I3 $C(1) = C(1) - D(1) \Rightarrow C(1+1)/8(1)$ $30 = CNT I = 1^{-1} + MM1$ I = N - I3 $C(1) = C(1) - D(1) \Rightarrow C(1+1)/8(1)$ $30 = CNT I = 1^{-1} + MM1$ I = N - I3 $C(1) = C(1) - D(1) \Rightarrow C(1+1)/8(1)$ $30 = CNT I = 1^{-1} + MM1$ I = N - I3 $C(1) = C(N) - N(N) + D(NM1) + C(NM1) = 2^{-1} = 1^{-1} + 2^{-1} = 1^{-1$ $\begin{array}{l} B(N) = (Y(N) - Y(NML)) / D(NML) + D(NML) \Rightarrow (C(NML) + 2 \cdot C(N)) \\ DO = 0 I = 1 \cdot NML \\ B(I) = (Y(I+1) - Y(I)) / D(I) - D(I) \Rightarrow (C(I+1) + 2 \cdot C(I)) \\ D(I) = (C(I+1) - C(I)) / D(I) \\ C(I) = 3 \cdot C(I) \\ = 0 + 1 - C(I) - 0 + 1 \\ C(N) = 3 \cdot C(N) \\ D(N) = D(N-1) \\ RETURN \end{array}$ $\frac{RETURN}{50 B(I) = (Y(2) - Y(1))/(X(2) - X(1))}$ D(1)=0. D(1)=0. B(2)=B(1) C(2)=0. D(2)=0. RETURN SUBROUTINE FFT(KODE.N.DELTA.X.*) PROGRAM TO IMPLEMENT THE FAST FOURIER TRANSFORM WHEN THE NUMBER OF DATA POINTS IS AN INTEGER POWER OF TWO. TAKEN FROM PAGES 254-265 OF "METHODS OF DISCRFTE SIGNAL AND SYSTEM ANALYSIS", BY JONG. c Ĉ COMPLEX X(N), W, X1, CMPLX $\begin{array}{l} IA = C \\ N1 = N \\ S N2 = N1/2 \\ IF (N2 \approx 2 \cdot NE \cdot N1) & GOTO & 100 \\ IP = IP + 1 \\ N1 = N2 \\ IF (N1 \cdot GT \cdot 1) GOTO & S \\ PN = 5 \cdot 293195/N \\ L = N/2 \\ IR 1 = IR - 1 \\ K1 = 0 \\ DO & 3O & IS = 1 \cdot IR \\ IS & DO & 2O & I = 1 \cdot L \\ K = KI + I \\ KPL = K + L \\ AM = KB ITP (K1/2 \Leftrightarrow IR1 \cdot IR) \\ IF (AM - NE \cdot 0 \cdot) & GOTO & 13 \\ X1 = X(KPL) \\ GOTO & 1.9 \\ IS & ARG = AM \Leftrightarrow PN \\ C = COS(ARG) \\ S = -KOSF \Leftrightarrow SIN(ARG) \\ W = CMPLX(KPL) \\ X = W \Leftrightarrow X(KPL) \\ \end{array}$ I 2 = C

13 X(EPL)=X(K)-X1 X(K)=X(K)+X1 20 El=K1+1 K1=K1+L 1F(K1+ET+N) GOTO 15 K1=0 fR1 = fr1 - 130 L=L/2 30 40 K=1+N K1=K3173(K=1+T2)+1 IF(K1+UP+K) GOTO 40 X1=X(K) x1=x(k) x(k)=x(k1) x(k1)=k1 40 CONTINUE IF(DELTA-EQ.1.) RETURN 90 50 k=1.N 30 x(k)=DELTA=x(k) PETURN 00 UT((101) N 100 WRITE(6.101) N 101 FORMAT(//,* *** N= *.16,* IS NOT A POWER OF 2. FFT RUN ABORTED *** 1.1.1 RETURNI FND FUNCTION KBITR(K,IP) KBITR=0 K1=K K1=K DU 1 [=1,[? K2=K1/2 KBIT?≠2☆KBTT?+K1-2☆K2 1 KI=K2 RETURN END END DOUBLE PRECISION FUNCTION DFUN(DX) SUBPOUTINE TO FORM THE INTEGRAND IN GAUSSQ. SINCE GAUSSQ IS USED FOR NUMERICAL QUADRATURE WITH TWO DIFFERENT INTEGRANDS, WHICH INTEGRAND IS EVALUATED BY FUN IS DETERMINED BY THE FLAG IFUN, PASSED TO FUN BY THE MAIN PROGRAM THRU COMMON STORE. IMPLICIT COMPLEX*16 (C), REAL*8 (D) COMPLEX*16 CD/(.81D-3,0.DO)/,CONE/(1.D0,0.D0)/,CTWO/(2.D0,0.D0)/, ≎ĎС́мрі́х >DCMPLX COMPLEX CURNT REL #3 DMUSIG/21.8654800/.DHD3/1.59D-3/.DG/2.78D-3/.DCOMP(2). >D1474Y(100) FOUTVALENCE (CHR.DCOMP(1)) COMMON DAPAAY.CURNT.FUN.R.TWOPIE FORM COMPLEXES VERSION OF THE COMPLEX#8 CURRENT PASSED TO IFUN THRU THE COMMON VARIABLE CURNT CURNTD=DCMPLX(DBLE(REAL(CURNT)).DBLE(AIMAG(CURNT))) IF(DX -NE.0.DO) GOTO 10 LAMRDA=0: INTEGRAND=0 DFUN=0.DO RETURN c c C DFUN=0.DO RETURN EVALUAT: THE NUMERATOR (WITHOUT THE KERNEL) 10 DT1=DEXP(-DX¢DHD3) DT2=((DT1+1.DO)¢DT1+1.00)¢DEXP(-DX¢DG)¢DKPRLK(DX) IF FREQUENCY=0, ALMOST DONE IF(TWOPIF.NE. 0.) GOTO 20 FREQUENCY IS ZERO CHR=DCMPLX(DT2/2.DO.O.DO)¢CURNTD GOTO 30 WEITHER LAMBDA NOR FREQUENCY IS ZERO Ċ C С C MEITHER LAMBDA NOR FREQUENCY IS ZFR0
20 CSQ=CDSDRT(DCMPLX(DX¢DX,DBLE(TWOPIF)¢DMUSIG))
CT=CDEXP(CD¢CSQ)
CT=CONE/CT
CCOSH=(CT+CTI)/CTW0
CSINH=CCOSH-CTI
CRATID=DCMPLX(DX,0+D0)/CSD
CDENDM=CONE+(CCOSH+CRATID¢CSINH)/(CSINH/CRATID+CCOSH)
CHR=DCMPLX(DT2,0+D0)¢CURNTD/CDENOM
NUW CALCULATE INTEGRAND
30 DRTL=DX*DAUE(P)
DJ1=DJ1LK(DPTL)
IF(IFUN +E4+1) DFUN=DCOMP(1)¢DJ1¢DX
IF(IFUN +E4+2) DEUN=DCOMP(2)¢DJ1¢DX
RETUPN C RETURN FND

OF POOR OUALITY

٠.

DOUPLE PARCISION FUNCTION DEPREK(DLMBDA) SUBPRUGRAM TO EVALUATE THE FUNCTION EPRIME(LAMBDA). THE ARGUMEN DLMBDA MUST BE REALMA. USES THE RESULT THAT THE ANTIDEPIVATIVE DE ALLYN A W TE EDUAL TO С THE ARGUMENT DLMDDA MUST BE REALER. USES THE RESULT THAT THE ANTIDEPIVATIVE OF J1(X) \Leftrightarrow X IS EQUAL TO PI \Leftrightarrow X \Leftrightarrow (J1(X) \Leftrightarrow HO(X) - H1(X) \Leftrightarrow JO(X)) / 2 FOR ARGUMENTS DLMBDARX (LE, 8, THE DIFFERENCE IS EVALUATED DIRECTLY. FOP DLMBDARX (GT, 8, SOME SIMPLIFICATION IS UTILIZED (SEE THE PAPER DATED JULY 23). REF: "HIGHER TRANSCENDENTAL FUNCTIONS", VOL. 2, P. B9, EQUATION (5) WITH NUEL PEQUIPES FUNCTION SUBPROGRAMS DHOLK(DX), DHILK(DX), DAOSM(DX), DAISM(DX), AND THE REALER IMSL SUBROUTINE MMBSJN. MODIFIED SEPT 19 85 AT WSU TO PEPLACE MMBSJN WITH MY E.P. FUNCTIONS DJOLK AND DJILK. С DJOLK AND DJILK. IMPLICIT RF4L\$A (D) PE4L\$A DPI/3+14159265353979/,DR1/+003175D0/+DR2/+0254D0/,DJU(2), \$DJU(2) ODICE/ DATA [FLAG/+1/ CHSCK TO STE IF POUTINE HAS BEEN CALLED BEFORE IF (IFLAG +NE+ -1) GOTO 10 NO PREVIOUS CALL - PEPFURM INITIALIZATION C С DCONST=3.DO&DPI/(DP2-DP1) RESET FLAG 50 INITIALIZATION WON'T BE PERFORMED AGAIN IFLAG=-1 C CALCULATE KPRIME(LAMBDA) C 10 DKPRLK="+DD CHECK TO SEE IF ARGUMENT = 0 IF(DLMBDA +EQ+ D+DD) RETURN DEGIN BY FORMING UPPER AND LOWER INTEGRATION LIMITS C С DU=D#200LM3DA DU-DH20DCH3DA DL=DP10DLM3DA FORM BESSEL FUNCTIONS OF THE FIRST KIND AT THE UPPER AND LOWER INTEGRATION LIMITS DJU(1)=DJOLK(DU) DJU(2)=DJOLK(DU) ſ DJL(1)=0J0LK(DL) DJU(2)=0J1UK(DU)FORM DIFFERENCE AT THE UPPER INTEGRATION LIMIT DECIDE WHICH TECHNIQUE TO USE TO EVALUATE J1(DU)#H0(DU)-H1(DU)#J0(DU) IF (DU .GT. 8.DU) G0TU 20 DIFFU=DJU(2)#DH0LK(DU)-DHLLK(DU)#DJU(1) ٢ DIFFU=DJU(2)*DHOLK(DU)~DHLLK(DU)*DJU(1) GOTO JO DIFFU=DJU(2)*DA05M(DU)~DA15M(DU)*DJU(1) + 2.D0/(DPI*DU) FOFM DIFFERENCE AT THE LOWER INTEGRATION LIMIT DECIDE WHICH TECHNIQUE TO USE TO EVALUATE J1(DL)*HO(DL)-H1(DL)*JO(DL) 30 IF (DL .6T. 8.D0) SOTO 40 DIFFL=DJL(2)*DHOLK(DL)~DH1LK(DL)*DJL(1) GOTO 50 40 DIFFL=DJL(2)*DA05M(DL)~DH1LK(DL)*DJL(1) + 2.D0/(DPI*DL) EVALUATE KPRIME 50 DKPRLK=DCONST*(DR2*DIFFU-DR1*DIFFL)/DLMBDA RETURN END FUNCTION DHOLK(DX) С END FUNCTION DHOLK(DX) POUGLE PRECISION SUBPROGRAM TO EVALUATE THE STRUVE FUNCTION OF DROER ZERO FOR DOUBLE PRECISION ARGUMENTS DX «LE« 8 USING A SUMMATION OF WEIGHTED CHEBYSHEV POLYNOMIALS« REQUIRES THE SUBROUTINE DCOPS (NASA LEWIS COMPUTER LIBRARY) TO FORM AND SUM THE POLYNOMIALS« REF: "THE SPECIAL FUNCTIONS AND THEIR APPROXIMATIONS", VOL« 2, P. 370, BY YUDELL LUKE« С ç INPLICIT RFALMA (D) REALMA R(L)/1.00215843609712.-1.63969292681309,1.50236939618293, -=.72451153021218,.189533273710931.-.03067052022988, -=.337.3514473751940 -5,-26.9550143126020-5,1.63746169261230-5, -=.7824.44085082540-10,302.15931841530-10,-9.632664495D-10, -=.25773370890-10,-539.53949D-15,11.58332D-15/ CHECK FUR ARGUMENT 0UT OF RANGE IF(DARS(DX) .GT. 8.00) RETUPN FURM THE ARGUMENT FOR THE CHEBYSHEV POLYNOMIALS DXD8=DX/8.00 DARG=2.0000XD800XD8-1.00 SUM THE POLYNOMIALS CALL DCNPS(UHOLK,DARG,8,15,99) COMPLETE THE SYALUATION AND RETURN DMOLK=DMOLK0XDA C C C C DHOLK=DHOLK¢DXDA 19 RETURN END

```
SUBROUTINE DONPS(Y,X,C,N,*)
           DIMENSION C(N)
DOUBLE PRECISION C+Y+X+HO+HI+H2+ARG
   TEST OF DIMENSION
IF(DABS(X) +GT+ 1+DD)RETURN1
IF(N)1+1+2
1 RETURN
    2 IF(N-2)3+4+4
3 Y=C(1)
RETURN
    INITIALIZATION
4 ARG=X+X
            H1=0.00
            H)=0+00
            DO 5 I=1+N
K=N+I
            H2=H1
           H1=H0
         H0=ARG#H1-H2+C(K+1)
Y=0.5D0%(C(1)-H2+H0)
RETURN
            FND
 END
FUNCTION DHILK(DX)
DOUBLE PRECISION SUBPROGRAM TO EVALUATE THE STRUVE FUNCTION OF
ORDER ONE FOR DOUBLE PRECISION ARGUMENTS DX .LE. B USING A
SUMMATION OF WEIGHTED CHEBYSHEV POLYNOMIALS.
PEQUIRES THE SUBROUTINE DCNPS (NASA LEWIS COMPUTER LIBRARY) TO
FORM AND SUM THE POLYNOMIALS.
REF: "THE SPECIAL FUNCTIONS AND THEIR APPROXIMATIONS",
VOL. 2, P. 370, BY YUDELL LUKE.
END
FUNCTION DAOSM(DX)
DOUBLE PRECISION SUBPROGRAM TO EVALUATE THE SUMMATION ASSOCIATED
WITH THE STRUVE FUNCTION OF ORDER ZERO FOR DOUBLE PRECISION ARGU-
MENTS DX .GE. A USING A SUMMATION OF WEIGHTED CHEBYSHEV POLYNOMIALS.
Requires the subroutine DCNPS (NASA LEWIS COMPUTER LIBRARY) TO
FORM AND SUM THE POLYNOMIALS.
REF: "THE SPECIAL FUNCTIONS AND THEIR APPROXIMATIONS",
VOL. 2, P. 371, BY YUDELL LUKE.
        IMPLICIT REAL#8 (D)

REAL#3 D(20)/.972337275764239,-696.891281138625D-5,

#13.2051037870371D-5,-1.06325825284416D-5,9819.3294286525D-10,

=1223.0545444377D-10,139.40833118D-10,-34.4358225604D+10,

=1223.0545444377D-10,-1.6288744137D-10,-465680728D-10,

=.1091504796D-10,3120.05243D-15,-942.0207D-15,298.47947D-15,

=.38.724160-15,33.33712D-15,-12.0798D-15,4.43821D-15,-1.67859D-15/

REAL#3 DPID2/1.5707963267349/
REALER UP(02/1-5/0/06/526/949/
CHECK FOR ARGUMENT OUT OF PANGE
IF(DANS(DX) +T+ 8+00) RETURN
FORM THE ARGUMENT FOR THE CHERYSHEV POLYNOMIALS
DADX=8+00/0X
DARG=2+D0@DADX@DADX=1+00
SUM THE POLYNOMIALS
CALL DCNPS(DAOSM+D42G+D+20+99)
COMPLETE THE FVALUATION AND RETURN
DAOSM=DAOSM/(DPID2@DX)
99 RETURN
 99 RETURN
            END
```

C

С

C C

C

CCCCCCCC

c c

C

C

CCCCCC

C

C

с с

FUNCTION DAISM(DX) BOUBLE PRECISION SUBPROGRAM TO EVALUATE THE SUMMATION ASSOCIATED WITH THE STRUVE FUNCTION OF ORDER ONE FOR DUTALE PRECISION ARGU-MENTS DX «GF« & USING A SUMMATION OF WEIGHTED CHEBYSHEV POLYNOMIALS« REQUIRES THE SUBROUTINE DENPS (NASA LEWIS COMPUTER LIBRARY) TO FORM AND SUM THE POLYNOMIALS« REF: MTHE SPECIAL FUNCTIONS AND THEIR APPROXIMATIONSM« VOL« 2, P« 371, RY YUDELL LUKE» č C C C C C C IMPLICIT REAL#A (D)
REAL#A E(17)/1.00157647293866,750.3160512432570-5,
\$\$\phi - 7.04393326451905D-5,266205393382266D-5,-1884.1157753405D-10,
\$\$104.9014958394D-10,-26.1261989905D-10,4.2362690104D-10,
\$\$\phi - 7955155531D-10,.1679973006D-10,-3907.13821D-15,985.4309D-15,
\$\$\phi - 266.35794D-15,76.45035D-15,-23.12961D-15,7.33212D-15,
\$\$\phi - 2.62334D-15/
REAL#A DPID2/1.5707963267949/
CHECK FOP ARGUMENT OUT OF PANGE
IF(DAB5(DX) -LT. 3.DD) RETURN
FOPM THE ARGUMENT FOR THE CHEBYSHEV POLYNOMIALS
D3DX=3.D0/DX
DARG=2.D0#DADX#DBDX-1.D0
SUM THE POLYNOMIALS
CALL UCNPS(DA15M,DARG,E,17,99)
COMPLETE THE EVALUATION AND RETURN
DA15M=DA15M/DPID2
P9 RETURN ٢ C C C 19 RETURN END DOUBLE PRECISION FUNCTION DJILK(DX) DOUBLE PRECISION SUBPROGRAM TO EVALUATE THE BESSEL FUNCTION OF THE FIRST KIND OF OPDER 1 AT THE DOUBLE PRECISION ARGUMENT DX. FOR ARGUMENTS DABS(DX) .LE. 8 A TRUNCATED JACORI SERIES OF CHERYSHEV PULYNOMIALS IS USED (SEE THE FIRST REFERENCE BELOW). FOR ARGUMENTS DABS(DX).GT. A A TRUNCATED ASYMPTOTIC SERIES IS USED (SEE SECOND PEFERENCE BELOW). REQUIRES THE SUBROUTINE DONPS (NASA LEWIS COMPUTER LIBRARY) TO FORM AND SUM THE CHEBYSHEV POLYNOMIALS. REF: "THE SPECIAL FUNCTIONS AND THEIR APPROXIMATIONS", VOL. 2, P. 332, BY YUDELL LUKE. PEF: "CHERYSHEV SERIES FOR MATHEMATICAL FUNCTIONS", NAT. PHYS. LAB. MATH. TABLES, VOL V, P. 33, BY C.W. CLENSHAW. END c CCCCCCCC ARGUMENTS IMPLICIT REAL®S (D) REAL#3 08(15)/.54335877060526400,-1.1918011605412200. $\begin{array}{l} \texttt{A} \in \texttt{A} \subseteq \texttt{$ С C 03PI04=3.0000PI(4.00 09PI04=3.0000PI(4.00 0PI(1)=0PI(1)/2.00 091(1)=001(1)/2.00 IFLAG=1 IFLANTI 5 JJLK=0.DO CHECK FOR ARGUMENT EQUAL TO ZERO IF(DX *EQ: 0.DO) RETURN DECIDE IF JACOBI OR ASYMPTOTIC SERIES WILL BE USED IF(DA35(DX) *GT* 8.DO) GOTO 10 USE JACOBI SERIES FORM THE ARGUMENT FOR THE CHERYSHEV POLYNOMIALS PORTO С С HURM THE ARGUMENT FOR THE CHERYSH DXD8=DX/8.00 DARG=2.D00DXD80DXD8-1.D0 SUM THE POLYNOMIALS CALL DCNPS(DJ1LK,DAPG,D8,15) FINISH THE CALCULATION AND RETURN DJ1LK=DXD80DJ1LK 99 RETURN č C C RETURN NODIFIED ASYMPTOTIC SERIES D3DX=8+D0/DX D4RG=2+D0¢D8DX≎D8DX-1+D0 USE C 10

SUM THE POLYNUMIALS FOR PI(X) Call DCNPS(0P,Darg,DP1+15) SUM THE POLYNUMIALS FOR Q1(X) C C CALL DCVPS(00,DAPG,D01,15) D0=D000ADX FDVM THE ASYMPTOTIC EXPRESSION DJ1LK=(DCONST#(DP#DCOS(DX-D3PID4)-DQ#DSTN(DX-D3PID4))/DSQRT(DX) C RETURN END DOUBLE PRECISION FUNCTION DJOLK(DX) DOUBLE PRECISION SUBPRUGRAM TO EVALUATE THE BESSEL FUNCTION OF THE FIRST KIND OF ORDER O AT THE DOUBLE PRECISION ARGUMENT DX. FOR ARGUMENTS DABS(DX).LT. A A TRUNCATED JACOBI SERIES OF CHEBYSHEV POLYNOMIALS IS USED (SEE THE FIRST REFERENCE BELOW). FOR ARGUMENTS C ç DANS(DX) GE. & A TPUNCATED ASYMPTOTIC SERIES IS USED (SEE SECOND AND THIRD REFERENCE BELOW). REPUTRES THE SUBPOUTINE DONPS (NASA LEWIS COMPUTER LIBRAPY) TO FORM AND SUM THE CHEBYSHEV POLYNOMIALS. PEE: "THE SPECIAL FUNCTIONS AND THEIR APPROXIMATIONS", VOL. 2, P. 331, AY YUDELL LUKE. REF: "CHEBYSHEV SERIES FOR MATHEMATICAL FUNCTIONS", NAT. PHYS. LAB. MATH. TABLES, VOL V, P. 32, BY C.W. CLENSHAW. C IMPLICIT REAL®A (b) REAL®B DB(14)/+157727471474890D0,-872.344235285222D-5, © 26517861320335D0,-370094993872649D0,+158067102332097D0, © -348457674114089D-5,431.318006945760D-5,-46.0626166206275D-5, © 3.24603288310051D-5,-17619.4690776215D-10,750.8163592419D-10, © -26.7725353056D-10,78436.97314D-15,-1943.83469D-15, © 41.25321D-15,-758855.D-20,1222.0-20,-17.D-20/, © PDC(14)/1.99892069869504D0,-53.6522046813212D-5, © 30751.8478751947D-10,-517.0594537606D-10,16.3064646352D-10, © -78640.91377D-15,5168.26239D-15,-430.45789D-15,43.26596D-15, © -78640.91377D-15,5168.26239D-15,-430.45789D-15,43.26596D-15, © -78640.91377D-15,5168.26239D-15,-630.45789D-15,43.26596D-15, © -7414.479410605D-10,179.7245724797D-10,-7.2719159369D-10, © 42201.21905D-15,-3206.74742D-15,300.61451D-15,-33.36328D-15, © 4.25532D-15,-60999.D-20,9662.D-20,-1669.D-20,311.D-20/ DATA DPI/3.14159265358979/,IFLAG/-1/ C MFCK TO SEE IF ROUTINE MAS BEEN CALLED BEFORE IF(IFLAG.NE.-1) GOTO 5 C PEHFORM INITIALIZATION AND RESET FLAG DCONST=DSQRT(2.D0/DPI) IF(IFLAG •NE • - i) GOTO S
PERFORM INITIALIZATION AND RESET FLAG
DCONST=DSQRT(2.D0/DP1)
DPID4=DPI/4.D0
DPC(1)=DPO(1)/2.D0
IFLA*=1
5 DJOLK=1.D0
CMFCK FOH ARGUMENT EQUAL TO ZFPO
IF(DX •F0+ 0.D0) RETURN
2FCIDE IF JACOPE OR ASYMPTUTIC SERIES WILL BE USED
IF(DANS(DX) •GT • 0.D0) GOTO 10
USE JACOPE OR THE CHEBYSHEV POLYNOMIALS
DXD8=DX/0.D0
DAPG=2.D00DNX00DA=1.D0
SUM THE POLYNOMIALS
CALL DCNPS(DJOLK,DARG.DB,10)
49 RETURN
USE MODIFIED ASYMPTOTIC SERIES
10 D0DX 2000 ADX=1.D0
SUM THE POLYNOMIALS FOR PO(X)
CALL DCNPS(DP,DARG.DD,14)
D0=D20DDX
NOW FORM ASYMPTOTIC APPROXIMATION TO JO(X)
DJCLK=(DCONST@(DP0COS(DX-DPID4)-DQ@DSIN(DX-DPID4))/DSQRT(DX)
RETURN
FND C С C C C С C С RETURN END DUBLE PRECISION FUNCTION BECT(I) FUNCTION TO EVALUATE THE INDIVIDUAL TEAMS IN THE SERIES OF MAGNETIC FLEED INTENSITIES (EITHER ROLL OF IMAGINARY) ARISING FROM THE INVERSE MANKEL INTEGRATION DATWEEN THE LIMITS OF LAMBDA SUB(I) AND LAMBDA SUB(I+1). SEE THE PAPER DATED JULY 25. С IMPLICIT REAL®S (D Real®s Dapray(100) Commun Darpay DECT=DAPRAY(I) RETURN FND

SUBRINTING BREUL(FCT, SUM, MAX, EPS, IER, N) COPIED FULFR CONVERGENCE ACCELERATION SUBROUTINE DIMENSION Y(IS) DOUBLE PRECISION FCT, SUM, Y, AMN, AMP IF (MAX)1,1,2 1 IER=-1 GUTO 12 INITIALIZE EULER TRANSFURMATION 2 IER=1 [=1 M=1 N=1 Y(1)=FCT(N) SUM=Y(1)↔SDO START EULER LOOP 3 J=0 4 I=I+1 IF(I-MAX)5,5,12 5 N=1 AMN=FCT(I) D0 6 K=1,M AMP=(AMN+Y(K))\$.500 AMP=(AMN+Y(K)) ↔ 5D0 Y(K)=AMN 5 AMN=AMP CHECK EULER TRANSFORMATION IF(DABS(AMN)=DABS(Y(M)))7,9,9 7 IF(M=15)8,9,9 9 M=M+1 Y(M)=AMN AMN=-5D0%AMN UPDATE SUM 9 SUM=SUM+AMN IF(AB5(SNGL(AMN))=EPS⇔ABS(SNGL(SUM)))10,10,3 TEST END OF PROCEDURE 10 J=J+1 C 10 J=J+1 IF(J-5)4+11+11 11 IER=0 12 RETURN END SUBROUTINE QGAUSS (DLOWER, DUPPER, DFUN, DPELER, DABSER, DINT, DERR, SUBRIOUTINE OUR STUDIED ALLOWING TO CARE SHEET THE THE THE THE ALG.
SUBROUTINE TO ESTIMATE THE INFEGRAL OF THE FUNCTION DEUN BETWEEN THE LIMITS DEDWER AND DUPPER.
PARAMETERS DRELFR AND DABSER ARE THE USER REDUESTED RELATIVE AND ABSOLUTE ERRORS IN THE ESTIMATED VALUE OF THE INTEGRAL DINT. IF DABSER=0, UNLY THE RELATIVE ERROR CRITERION OR THE RELATIVE ERROR CRITERION. HAS BEEN ESTIMATED TO BE ACHIEVED.
PARAMETER DERR IS THE RETURNED ESTIMATE OF THE ABSOLUTE ERROR IN DINT. PARAMETER IFLAG IS SET TO ZERO IF NO CONVERGENCE IS OBTAINED; OTHERWISE, IFLAG INDICATES THE NUMBER OF INTEGRAL ESTIMATES CALCULATED MINUS 1.
SUBPROGRAMS REQUIRED - DEUN, DEGUAD, CONVERT
HEF: "GAUSSIAN QUADRATURE FURMULAS", BY STROUD AND SECREST (THE FUNCTION APGUMENTS AND THEIR WEIGHTS WERE TAKEN FROM THIS BOOK.) IFLAG) ¢ THIS BOOK.)

Ĉ

С

С

C

С

C

DATA 0;,W32/*7018610009470100-2;*1627439473090570-1; 0*253920653092;210-1;*3427386291302140-1;*4283589602222670-1; 0*5093405926237520-1;*5863409347853550-1;*5582222277636180-1; 0*7234579413848350-1;*7819349578707030-1;*4331192422694680-1; 0*576309300440380-1;*9117387869576390-1;*9384439908080460-1; 0*9561372007327490-1;*957037637472780-1/ 0474 0;X40/*998237703710559;*990726238699457; 0*97725936433774;*957916319213792;*932812808278677; 0*97098806958874;**957916319213792;*932812808278677; 0*97098806958874;**95595950312260;*824612230833312; С C C IFLAG=1 DINT=DINT15 RETURN NO CUNVERGENCE - CALCULATE THE 24 POINT INTEGRAL ESTIMATE 10 DINT24=DGOUAD(DLOWER,DUPPER,DEUN,12,DGX24,DGW24) DETERMINE WHETHER CONVERGENCE WAS OBTAINED CALL CONVT(DINT16,DINT24,DRELER,DARSER,DERR,ICONV) IF(ICONV +NE+ 1) GOTO 20 CONVERGENCE - SET UP VALUES, AND RETURN Ĉ C 1 . C IFLAG=2 DINT=DINT24 RETURN C NO CONVERGENCE - CALCULATE THE 32 POINT INTEGRAL ESTIMATE 20 DINT 32=DGQUAD(DLOWER, DUPPER, DEUN, 16, DGX 32, DGW 32) DETERMINE WHETHER CONVERGENCE WAS OBTAINED CALL CONVT(DINT24, DINT32, DRELER, DABSER, DERR, ICONV) IF(ICONV +NE+ 1) GOTO 30 CONVERGENCE - SET UP VALUES, AND PETURN č ۲ С 18144=3 DINT=DINT32 RETUPN c c NO CONVERGENCE - CALCULATE THE 40 POINT INTEGRAL ESTIMATE 30 DINT4D=0GJUAD(DLOWER, DUPPER, DEUN, 20, DGX40, DGW40) DETERMINE WHETHER CONVERGENCE WAS OBTAINED CALL CONVT(DINT32, DINT40, DRELER, DABSER, DERR, ICONV) IF(ICONV +NE+ 1) GOTO 40 CDNVERGENCE - SET UP VALUES, AND RETURN TELEDE-C С IFLAG=4 DINT=DINT40 1 RETURN c NO CONVERSENCE 40 IFLAGED DINT=DINT40 RETURN RETURN END SUBROUTINE CONVECTOR, DREW, DRELER, DABSER, DERR, ICONV SUBROUTINE TO TEST CONVERGENCE FOR THE SUBROUTINE REAUSS. DO INNEW ARE THE PREVIOUS AND PRESENT ESTIMATES OF THE INTEGRAL. PEE: MY PAPER DATED AUG 22. DOLD AND Ê IMPLICIT REAL®S (D) IMPLICIT PEAL®S (D) ICONVEL DERREDADS(DNEW-DOLD) IF THE PRESENT INTEGRAL ESTIMATE IS EQUAL TO ZERO, PROCEED TO TH ABSOLUTE ERROP TEST (AFTER MAKING SURE THAT DABSER .NE. O). IF(DMEW .EQ. 0.00) GOTO 10 ESTIMATE NOT EQUAL TO ZERO - CHECK THE RELATIVE ERROR CRITERION IF(DERR/DARS(DNEW) .LF. DRELER) RETURN NO RELATIVE FAROR CONVERGENCE - IF DABSER=0, THEN THERE IS NO CONVERGENCE. 10 IF(DARSER .EQ. 0.00) GOTO 20 CHECK THE ABSOLUTE ERROR CRITERION. TF(DERR .LE. DABSER) RETURN PROCEED TO THE Ĉ C C C

APPEHDIX E

er i

FORTRAN SOURCE CODE FOR CALCULATING FORCE AT TIME T=50 MICROSECONDS

FILE: FORMU FORTRAN VIAL PROGRAM TO ESTIMATE THE TOTAL FORCE PETWOON THE COIL AND THE SKIN BY CALCULATING (10407)/4 TIMES THE ESTIMATE OF THE INTEGRAL OF BROBROR - BZOSZOH BETWEEN R=-01 AND R=1.0 INCHES. REF: MY PAPER DATED JAN 13 *86. ç ć REAL BP50N(11)/.017014,.306449,.450552,.503336,.461798,.233143, *.106437,.046160,.325120,.014479,.008452/ REAL PZ50N(11)/.224464,.213477,.142827,.082718,.033232,...31324, *.047354,...017196,...006434,...02941,...01629/ PEAL BR50NS(11),CR50NS(11),DR50NS(11),3250NS(11),CZ50NS(11), SQUARE THE CALCULATED B FIELDS DU 10 I=1,11 BR50N(I)=BR50N(I) 10 BZ50N(I)=BZ50N(I) CHANGE THE RADII AT WHICH THE B FIELDS WERE CALCULATED FROM INCHES TO METERS TO METERS С c TO METERS DO 15 I=1,11 PR(I)=RR(I)*TOMTRS 15 RZ(I)=RZ(I)*TOMTRS 15 RZ(I)=RZ(I)*TOMTRS FIT SPLINE FUNCTIONS TO BR#BR AND BZ#BZ CALL SPLINE(N,RZ,RZ50N, BR50NS, CZ50NS, DZ50NS) CALL SPLINE(N,RZ,RZ50N, RZ50NS, CZ50NS, DZ50NS) C C PEPFORM THE INTEGRATIONS č RINTSO=0. ZINTSO=0. LCCP TO ADD THE CONTRIBUTIONS TO THE INTEGRAL BETWEEN KNOTS DO 20 I=1,10 C TP1=T+1 PINT50=RINT50+SPLINT(I,RR(IP1),RR(T),3350N(I),BR50NS(I),CR50NS(I), http://www.sciller.com/clines/cl C DONE EXCEPT FOR MULTIPLICATION BY CONSTANT + 30 4 E T = CONST¢(RINT 50 - ZINT 50) F 50 L BS = F 50 MET 7 TOL B 5 WRITE (6, 1000) MONTH, DATE, F 50 MET, F 50 L BS 1000 FORMAT (*100 TPUT FROM FOR WU FORTRAN V1.1*, 10X, 44, 44, //* TOT AL * AT 50 USEC IS *, F 9.4, * NEWTON 5*, /, T25, *= *, F 9.4, * POUND 5*, //) STOP END FORCE END FUNCTION SPLINT(I, PUP, RLO, F, P, C, D) FORTTAN SUBPROGRAM TO EVALUATE THE INTEGRAL OF F(R) * R OVER THE INTERVAL RLO +LE+ R +LE+ RUP WITH I = SPLINE FUNCTION INDEX CORPESPONDING TO THIS INTERVAL PUP = UPPER INTEGRATION LIMIT (A KNOT) RLO = LOWER INTEGRATION LIMIT (A KNOT) C Ĉ C F = F(PLO) F =C C AL. Ref: My Poper Dated Jan 14 86.

PL01=PL0PL0 RL04=RL02#RL0 RL04=RL03#PL0 RL05=PL04#RL0 RU07=RU04#RL0 RU07=RU04#UP ORIGINAL PAGE IS OF POOR QUALITY RUPP=RUP0+UP PUP3=RUP2+RUP2+RUP RUP5=RUP4+RUP RUP5=RUP4+RUP CALCULATE FIPST TERM SPLINT=(((D\$RL0-C)*RL0+B)*RL0-F)*(PL02-3UP2)/2+ ADD SECOND TEPM SPLINT=SPLINT+((3+#D*RL0-2+*C)*RL0+B)*(RUP3-RL03)/3+ SPLINT=SPLINT+((3+#D*RL0-2+*C)*RL0+B)*(RUP3-RL03)/3+ SPLINT=SPLINT+(3+#D*RL0-2+*C)*RL0+B)*(RUP3-RL03)/3+ SPLINT=SPLINT+(3+#D*RL0-2+*C)*RL0+B)/4+ 1 C C ADD THIRD TERM SPLINT=SPLINT-(3.¢D¢RLO-C)¢(RUP4-RLO4)/4. ADD FOUATH TERM; DONE SPLINT=SPLINT+D¢(RUPS-RLOF)/3. RETUAN FND ٤. SUBPOUTINE SPLINE(N, K, Y, B, C, D) CCPIED SUBROUTINE SPLINE FOLLOWS INTEGER N REAL X(N), Y(N), B(N), C(N), C(N) INTEGER NM1, IB, I C INTEGEM NM1; IB; I PEAL T NM1=N-1 IF (N.LT.2)RETURN IF (N.LT.3)GO TO 50 D(1)=X(2)-Y(1) C(2)=(Y(2)-Y(1))/D(1) DO 10 I = 2; NM1 D(I) = X(I+1)-X(I) R(I) = 2*(D(I-1)*D(I)) C(I+1) = (Y(I+1)-Y(I))/D(I) C(I) = C(I+1)-C(I) 10 CONTINUE B(1)=-D(1) S(N)=-F(N-1) C(1)=0. C(N)=0. $\begin{array}{l} \widehat{C}(1) = 0 \\ C(N) = 0 \\ IF(N \in \overline{D} \circ 3) & GO TO 15 \\ C(1) = C(3)/(X(4) - X(2)) - C(2)/(X(3) - X(1)) \\ C(N) = C(N - 1)/(X(N) - X(N - 2)) - C(N - 2)/(X(N - 1) - X(N - 3)) \\ C(N) = -C(N)^{2} O(N - 1)^{4} C/(X(4) - X(1)) \\ C(N) = -C(N)^{2} O(N - 1)^{4} C/(X(4) - X(N) - X(N - 3)) \\ C(N) = -C(N)^{2} O(N - 1)^{4} C/(X(N) - X(N - 3)) \\ 15 DO 2G I = 2 \\ N \\ T = D(I - 1)/R(I - 1) \\ P(I) = B(I) - T^{4} D(I - 1) \\ C(I) = C(I) - T^{4} C(I - 1) \\ C(I) = C(I) - T^{4} C(I - 1) \\ \end{array}$ 20 CONTINUE C(N)=C(N)/B(N) D() 30 IE= 1, 341 D(N)=D(N-1) RETURN 50 B(I)=(Y(2)-Y(1))/(X(2)-X(1)) C(1)=0. D(1)=0. B(2)=0. B(2)=0. D(2)=0. RETURN **FND**

APPENDIX F

NOTATION AND LIST OF SYMBOLS

Unless stated otherwise, the rationalized MKS system is used in all equations and calculations. Equations are numbered consecutively, beginning with (1) in each chapter. Only those equations that are referenced in the text are numbered. Figures are identified by two alphanumeric characters. The first character is a number, indicating the number of the chapter in which the figure appears. The second character is a letter, alphabetically identifying the order in which the figures appear.

A chapter by chapter summary of the symbols used in this dissertation, listed in the order of their appearance, follows.

Chapter One

(س)Ζ	Terminal coil impedance (page 7)
ധ	Radian frequency (Fourier transform) variable (page 7)
i(t)	Time domain coil current (page 7)
I(ω)	Fourier transform of i(t) (page 7)
E _q (z,r,	 Azimuthal real space component of the electric in- tensity (page 7)
z	Axial coordinate in cylindrical coordinate system (page 7)
r	Radial coordinate in cylindrical coordinate system (page 7)

- H.(z,r,t) Radial real space component of the magnetic intensity (page 7)
- H.(z,r,t) Axial real space component of the magnetic intensity (page 7)
- f(t) Separation force magnitude between the coil and metal target (page 7)
- Impulse delivered to metal target by the coil (page 7)

Chapter Three

.

R ₁	Inner coil radius (page 12)
Re	Outer coil radius (page 12)
h	Coil axial width (page 12)
đ	Metal target thickness (page 12)
ቀ .	Azimuthal coordinate in cylindrical coordinate system (page 12)
ሥ	Permeability of free space (all materials considered in this dissertation are non-magnetic) (page 14)
a -	Conductivity of metal target (page 14)
j	Principal square root of -1 (page 15)
J _n (x)	Bessel function of the first kind of order n (page 15)
λ	Hankel transform variable (page 15)
Ŷ	Total Fourier-Hankel space transmission line voltage (page 16)
Î	Total Fourier-Hankel space transmission line current (page 16)
ጸ	Fourier-Hankel space transmission line complex propaga- gation coefficient (page 17)
î	Fourier-Hankel space transmission line phasor current corresponding to propagation in the direction of in- creasing z (page 17)
î	Fourier-Hankel space transmission line phasor current corresponding to propagation in the direction of de- creasing z (page 17)

- \hat{V}_{o-} Fourier-Hankel space transmission line phasor voltage corresponding to propagation in the direction of decreasing z (page 18)
- Z. Fourier-Hankel space transmission line characteristic impedance (page 19)
- **\widetilde{I}** Fourier space phasor current (page 19)

creasing z (page 18)

g Distance between metal target and the current sheet closest to target (page 20)

- z. Axial coordinate of current sheet closest to target (page 20)
- z, Axial coordinate of center current sheet (page 20)
- Z₂ Axial coordinate of current sheet furthest from target (page 20)
- Chapter Four

Ŷ..

- **V** Fourier space phasor voltage (page 23)
- Z. Fourier-Hankel space air transmission line characteristic impedance (page 26)
- X_a Fourier-Hankel space air transmission line complex propagation coefficient (page 26)
- Z. Fourier-Hankel space metal transmission line characteristic impedance (page 26)
- ۲ Fourier-Hankel space metal transmission line complex propagation coefficient (page 26)
- Z(0) Fourier-Hankel space impedance seen looking into metal transmission line at z=0 in Figure 3C (page 26)
- Physical Provide Provide
- PE Fourier-Hankel space voltage reflection coefficient
 (page 28)
- \widehat{V}_n Fourier-Hankel space total voltage at coil side face of metal target (page 30)

- Ŷ, Fourier-Hankel space total voltage on face of target opposite coil (page 30)
- î. Fourier-Hankel space total current at coil side face of metal target (page 30)
- î, Fourier-Hankel space total current on face of target opposite coil (page 30)
- F Vector force on metal target (page 32)

Permittivity of free space (all materials considered in this dissertation have a relative permittivity of 1) (page 32)

Chapter Five

ε

Γ1	Radius of inner loop of wire in field measuring plate (page 38)
re	Radius of outer loop of wire in field measuring plate (page 38)
h	Distance between tw o loops of wire of the same radius on front and back sides of field measuring plate (page 38)

Chapter Six

- Real part of the total coil impedance (page 47) R(w)
- Inductance of coil (page 47) L(ω)
- ٧. Initial voltage on energy storage capacitor prior to discharge (page 47)
- **ک** (دی) Dirac delta distribution (page 47)
- V. (ω) Fourier transform of energy storage capacitor voltage (page 47)
- С Capacitance of energy storage capacitor (page 48)
- Rec AC winding resistance of the coil (page 48)
- Real { } Symbol denoting "real part of $\{ \}$ " (page 48)

Instantaneous current in coil when clamp diode across Ι. energy storage capacitor begins conducting (page 51)

Appendix C

$H_n(x)$	Struve	function	of	order	n	(page	801
----------	--------	----------	----	-------	---	-------	-----

- T_n(x) Chebyshev polynomial of the first kind of order n (page 80)
- Y_n(x) Bessel function of the second kind of order n (page 81)

A_n(x) Partial sum associated with the Jacobi series expansion of $H_n(x) - Y_n(x)$ (page 81)

REFERENCES

- [1] R. D. Rudich, "Weather-Involved U.S. Air Carrier Accidents 1962

 1984 A Compendium and Brief Summary", American Institute of
 Aeronautics and Astronautics paper number AIAA-86-0327, January,
 1986.
- [2] British Patent Specification No. 505,433 issued to Rudolf Goldschmidt; May 5, 1939.

.

- [3] Dr. Robert Schrag, unpublished paper, "Experimental Study of the "Electro-Impulse Process by Magnetic Field Diagnostics", summarized in NASA CR 174919, May, 1985.
- [4] M.H.S. El-Markabi and E.M. Freeman, "Electromagnetic Properties of a Circular Cylindrical Coil in a Set of Planar Ferromagnetic Regions", IEE Proc., Vol. 129, Pt. A, No. 8, Nov., 1982.
- [5] General Electric Co., SCR Manual ~ Fifth Edition, Syracuse Park, N.Y., 1972.
- [6] Richard G. Hoft, ed., SCR Applications Handbook, International Rectifier Corp., El Segundo, Ca., 1977.
- [7] RCA Solid State Division, Solid State Devices Manual, Somerville, N.J., 1975.
- [8] Samuel Levy, "Electromagnetic Shielding Effect of an Infinite Plane Conducting Sheet Placed Between Circular Coaxial Coils", Proceedings of the Institute of Radio Engineers, Vol. 24, No. 6, June, 1936.
- (9) F.W. Grover, Inductance Calculations Working Formulae and Tables, Dover, 1962.
- [10] C.V. Dodd and W.E. Deeds, "Analytical Solutions to Eddy-Current Probe-Coil Problems", Journal of Applied Physics, Vol. 39, No. 6, May, 1968.
- [11] M. Once, "Analysis of Finite Solenoid Coil Near a Conductor", Electr. Eng. Jpn., 88, 1968.
- [12] Ian N. Sneddon, The Use of Integral Transforms, McGraw Hill Book Co., 1972.
- [13] Robert A. Chipman, Transmission Lines, McGraw Hill Book Co., 1968.
- [14] Robert Grover Brown, Robert A. Sharpe, William Lewis Hughes, and Robert E. Post, Lines, Waves, and Antennas: The Transmission of Electric Energy, 2nd. ed., Ronald Press, 1973.

- [15] Richard E. Matick, Transmission Lines for Digital and Communication Networks, McGraw Hill Book Co., 1969.
- [16] F.B. Hildebrand, Advanced Calculus for Applications, 2nd. edn., Prentice Hall, 1976.
- [17] Murray R. Spiegel, Mathematical Handbook of Formulas and Tables, McGraw Hill Book Co., 1968.
- [18] Arthur Erdelyi, ed., Higher Transcendental Functions, Vol. II, Bateman Manuscript Project, McGraw Hill Book Co., 1953.
- [19] G.N. Watson, A Treatise on the Theory of Bessel Functions, 2nd. edn., Cambridge University Press, 1980.
- [20] Milton Abramowitz and Irene A. Stegun, Handbook of Mathematical Functions with Formulas, Graphs, and Mathematical Tables, Dover Publications Inc., 1965.
- [21] Robert W. Hornbeck, Numerical Methods, Prentice Hall, 1975.
- [22] ACM Algorithms Distribution Service, IMSL Inc., Sixth Floor, NBC Building, 7500 Bellaire Boulevard, Houston, Texas, 77036.
- [23] International Mathematical and Statistical Library (IMSL) Inc., Sixth Floor, NBC Building, 7500 Bellaire Boulevard, Houston, Texas, 77036.
- [24] John R. Rice, Numerical Methods, Software, and Analysis, McGraw Hill Book Co., 1983.
- [25] George E. Forsythe, Michael A. Malcolm, and Cleve B. Moler, Computer Methods for Mathematical Computations, Prentice Hall, 1977.
- [26] Peter Henrici, Essentials of Numerical Analysis, John Wiley and Sons, 1982.
- [27] Avram Sidi, "Extrapolation Methods for Oscillatory Infinite Integrals", Journal of the Institute of Mathematics and its Applications, vol. 26, 1980.
- [28] David Levin and Avram Sidi, "Two New Classes of Nonlinear Transformations for Accelerating the Convergence of Infinite Integrals and Series", Applied Mathematics and Computation, vol. 9, 1981.
- [29] Avram Sidi, "The Numerical Evaluation of Very Oscillatory Infinite Integrals by Extrapolation", Mathematics of Computation, Vol. 38, No. 158, April 1982.
- [30] William F. Ford and Avram Sidi, "An Algorithm for a Generalization of the Richardson Extrapolation Process" (to be published).

[31] A. H. Stroud and D. Secrest, Gaussian Quadrature Formulas, Prentice Hall, 1966.

- [32] Personal conversation with Dr. Avram Sidi, The Technion, Israel Institute of Technology, Haifa, Israel.
- [33] J. H. Wilkinson, Rounding Errors in Algebraic Processes, Prentice Hall, 1963.
- [34] Frances B. Hildebrand, Introduction to Numerical Analysis, 2nd. edn., McGrav-Hill Book Co., 1974.
- [35] Mark Jong, Methods of Discrete Signal and System Analysis, Mc-Grav-Hill Book Co., 1982.
- [36] Janet P. Mason, A Fortran Subroutine to Evaluate Struve Functions for Integer Orders and Positive Real Arguments, NRL Memorandum Report 3181, Research Computation Center, Naval Research Laboratory, Washington, D.C., December 1975.
- [37] Yudell Luke, The Special Functions and their Approximations, Vol. II, Mathematics in Science and Engineering, Vol. 53-II, Academic Press, 1969.
- [38] Cornelius Lanczos, Applied Analysis, Prentice Hall, 1956.
- [39] C.W. Clenshaw, Mathematical Tables, Vol. 5, Chebyshev Series for Mathematical Functions, Her Majesty's Stationery Office, London, 1962.
- [40] John F. Hart et. al., Computer Approximations, John Wiley and Sons, 1968.
- [41] J.A. Stratton, Electromagnetic Theory, McGrav-Hill Book Co., 1941.
- [42] R.L. Schrag and G.W. Zumwalt, "Electro-Impulse Deicing: Concept and Electrodynamic Studies", American Institute of Aeronautics and Astronautics paper number 84-0021, January, 1984.
- [43] R.M. Bowley, P.J. King, G.J. Lewis, and I.J. Shellard, "Production of Short Mechanical Impulses by Means of Eddy Currents", Proc. IEE, Vol. 130, Pt. B, No. 6, Nov. 1983.
- [44] G.J. Lewis, "The Electrodynamic Operation of Electro-Impulse De-Icing Systems", American Institute of Aeronautics and Astronautics paper AIAA-86-0547, January, 1986.